

ISSN 1600-5767

Received 10 January 2022 Accepted 28 March 2022

Edited by S. Mühlbauer, Technical University of Munich, Germany

¹This article is part of a virtual special issue on Magnetic small-angle neutron scattering – from nanoscale magnetism to long-range magnetic structures.

Keywords: polarized neutron scattering; uniaxial polarization analysis; small-angle neutron scattering; micromagnetics; magnetic nanocomposites.



Uniaxial polarization analysis of bulk ferromagnets: theory and first experimental results¹

Artem Malyeyev,^a* Ivan Titov,^a Charles Dewhurst,^b Kiyonori Suzuki,^c Dirk Honecker^d* and Andreas Michels^a*

^aDepartment of Physics and Materials Science, University of Luxembourg, 162A Avenue de la Faïencerie, 1511, Luxembourg, ^bInstitut Laue–Langevin, 6 Rue Jules Horowitz, BP 156, F-38042 Grenoble Cedex 9, France, ^cDepartment of Materials Science and Engineering, Monash University, Clayton, Victoria 3800, Australia, and ^dISIS Neutron and Muon Source, Rutherford Appleton Laboratory, Didcot, United Kingdom. *Correspondence e-mail: artem.malyeyev@uni.lu, dirk.honecker@stfc.ac.uk, andreas.michels@uni.lu

On the basis of Brown's static equations of micromagnetics, the uniaxial polarization of the scattered neutron beam of a bulk magnetic material is computed. The approach considers a Hamiltonian that takes into account the isotropic exchange interaction, the antisymmetric Dzyaloshinskii–Moriya interaction, magnetic anisotropy, the dipole–dipole interaction and the effect of an applied magnetic field. In the high-field limit, the solutions for the magnetization Fourier components are used to obtain closed-form results for the spin-polarized small-angle neutron scattering (SANS) cross sections and the ensuing polarization. The theoretical expressions are compared with experimental data on a soft magnetic nanocrystalline alloy. The micromagnetic SANS theory provides a general framework for polarized real-space neutron methods, and it may open up a new avenue for magnetic neutron data analysis on magnetic microstructures.

1. Introduction

Polarized neutron scattering is one of the most powerful techniques for investigating the structure and dynamics of condensed matter, in particular magnetic materials and superconductors (Chatterji, 2006). Based on the seminal papers by Bloch (1936, 1937), Schwinger (1937) and Halpern & Johnson (1939), the theory of polarized neutron scattering was worked out in the early 1960s by Maleev et al. (1963) and Blume (1963). Several classic experimental studies (Shull et al., 1951; Moon et al., 1969; Rekveldt, 1971; Drabkin et al., 1972; Okorokov et al., 1978; Pynn et al., 1983; Mezei, 1986; Schärpf & Capellmann, 1993) have demonstrated the basic principles and paved the way for today's three-dimensional cryogenic polarization analysis device (CRYOPAD) (Tasset, 1989; Tasset et al., 1999; Brown et al., 1993; Okorokov & Runov, 2001). With this technique it becomes possible to measure 16 correlation functions [see the paper by Mezei (1986) for a detailed discussion], which provide important information on the nuclear and magnetic structure of materials [see Williams (1988) and Lovesey (1984) for textbook expositions of polarized neutron scattering].

However, for the scattering of cold (long-wavelength) neutrons along the forward direction – as implemented on a small-angle neutron scattering (SANS) instrument – it has only in recent years become possible to perform neutron polarization analysis (retaining the full two-dimensional scattering information) 'routinely': more specifically, uniaxial (also called longitudinal or one-dimensional) polarization analysis, where the polarization of the scattered neutrons is

analyzed along the direction of the initial polarization (Moon *et al.*, 1969). Clearly, this progress is due to the development of efficient ³He spin filters (*e.g.* Batz *et al.*, 2005; Petoukhov *et al.*, 2006; Okudaira *et al.*, 2020), which, in contrast to *e.g.* single-crystal analyzers, can be used over a rather broad wavelength range and cover a large detector acceptance angle. Note also that Niketic *et al.* (2015) and Quan *et al.* (2019*a,b*) report on the development of a novel neutron spin filter, based on the strong spin dependence of the neutron scattering on protons. For the combination of uniaxial polarization analysis with SANS, the term POLARIS has been coined (Wiedenmann, 2005). In contrast to CRYOPAD, which generally demands the sample to be in a zero magnetic field environment, POLARIS allows for the application of large magnetic fields.

The POLARIS method has been successfully employed for studying the superparamagnetic response of concentrated ferrofluids (Wiedenmann, 2005), proton domains in deuterated solutions (van den Brandt et al., 2006; Aswal et al., 2008; Noda et al., 2016), the multiferroic properties of HoMn₃ single crystals (Ueland et al., 2010), the role of nanoscale heterogeneities for the magnetostriction of Fe-Ga alloys (Mudivarthi et al., 2010; Laver et al., 2010), local weak ferromagnetism in BiFeO₃ (Ramazanoglu et al., 2011), nanometre-sized magnetic domains and coherent magnetization reversal in an exchange-bias system (Dufour et al., 2011), precipitates in Heusler-based alloys (Benacchio et al., 2019), the magnetic microstructure of nanoscaled bulk magnets (Honecker et al., 2010; Michels et al., 2012), the internal spin structure of nanoparticles (Krycka et al., 2010, 2014; Hasz et al., 2014; Grutter et al., 2017; Orue et al., 2018; Bender, Fork et al., 2018; Bender, Wetterskog et al., 2018; Oberdick et al., 2018; Ijiri et al., 2019; Bender et al., 2019; Honecker et al., 2020), and Invar alloys (Stewart et al., 2019). Polarization analysis further makes it possible to reveal the direction of the magnetic anisotropy in single-crystalline spin systems, e.g. an easy plane versus an easy axis anisotropy or the confinement of the propagation vector along certain crystallographic directions in chiral and other exotic magnets (Takagi et al., 2018; White et al., 2018). In all of the above-mentioned studies, the spinresolved SANS cross sections were obtained and analyzed, but the polarization of the scattered neutrons was not further investigated.

Historically, this is the domain of the neutron depolarization technique (see *e.g.* Halpern & Holstein, 1941; Hughes *et al.*, 1948, 1949; Burgy *et al.*, 1950; Maleev & Ruban, 1970; Rekveldt, 1971, 1973; Drabkin *et al.*, 1972; Maleev & Ruban, 1972; Rosman & Rekveldt, 1991, and references therein), where one measures the change in the polarization of a polarized neutron beam after transmission through a partially magnetized magnetic material. Analysis of the 3×3 depolarization matrix yields information on *e.g.* the average domain size and the domain magnetization. This type of polarization analysis on a SANS instrument has been termed 'vector analysis of polarization' by Okorokov & Runov (2001). Alternatively, it has been demonstrated that the neutron spinecho technique can resolve magnetic small-angle scattering (Grigoriev *et al.*, 2006; Rekveldt *et al.*, 2006). Spin-echo smallangle neutron scattering (SESANS) provides information on correlations on a length scale from about 10 nm to 10 μ m. In SESANS, the neutron spin precesses in a constant magnetic field and the neutron runtime difference due to sample scattering results in the dephasing of the neutron spins and in a loss of the measured beam polarization. Magnetic scattering can result in neutron spin-flip events that act as an additional optical element reversing the sense of the Larmor precession. The change of the neutron spin due to magnetic scattering can be exploited to study magnetic systems.

More recently, the method of polarized neutron dark-field contrast imaging (DFI) has been introduced for spatially resolved small-angle scattering studies of magnetic microstructures (Valsecchi et al., 2021). First experimental results on a sintered Nd-Fe-B magnet demonstrated not only that darkfield contrast from half-polarized SANS measurements can be observed but also that it becomes possible to separate and retrieve dark-field contrast for all spin-flip and non-spin-flip channels separately. The polarized DFI method has great potential for analyzing real-space magnetic correlations on a macroscopic length scale, well beyond what can be probed with a conventional SANS instrument. Similarly, first measurements of micrometre-sized magnetic correlations have been performed with an alternative neutron precession technique called spin-echo modulated small-angle neutron scattering (SEMSANS) (Li et al., 2021). With this setup, the spin manipulations are performed before the sample so that the measurement is not sensitive to large stray fields (related e.g. to the sample environment), and it even allows studies under beam depolarizing conditions. The polarization analyzer discriminates the polarization parallel to the analyzing direction, such that the scattering cross sections for the opposite neutron spin state are probed at the sample. The two-dimensional neutron polarization modulation observed on the detector is then integrated to yield the one-dimensional correlation function of the system.

For the above techniques, the analysis of magnetic materials is based on performing a neutron-spin analysis deliberately after the sample. The magnetic spin-flip scattering signal of the sample is utilized as a spin flipper to obtain exclusive sensitivity of the signal on local magnetization components. The projected correlation function is modified with the polarization of the scattered neutrons to reflect the additional phase in precession angles due to the spin-flip event.

In this work, we present a micromagnetic SANS theory for the uniaxial polarization of the scattered neutrons of bulk magnetic materials. The approach has recently been employed to analyze SANS cross sections directly in Fourier space (Michels, 2021) and is here extended to include the final polarization, which can be measured with a much higher precision than the individual cross sections (Brown, 2006). The continuum theory of micromagnetics allows one to characterize the large-scale magnetization distribution of polycrystalline magnets, which is determined *e.g.* by magnetic anisotropy and saturation-magnetization fluctuations, antisymmetric exchange, and dipolar stray fields. Since the validity of micromagnetic theory extends to the micrometre regime, the theoretical framework developed here may also serve as a basis for the above-mentioned polarized neutron techniques (SESANS, DFI, SEMSANS), which provide real-space information on large-scale magnetic correlations. Rekveldt *et al.* (2006) summarize the relevant expressions which relate the magnetization distribution of the material (obtained from micromagnetics) to the final polarization and the projected correlation function. The derived theoretical expressions for the polarization are tested against experimental SANS data on a soft magnetic nanocrystalline alloy.

The present paper is organized as follows: Section 2 summarizes the elementary equations of polarized neutron scattering, while Section 3 sketches the basic steps and ideas of the micromagnetic SANS theory. These two sections are well documented in the literature and may be skipped by the reader. They are included here merely to achieve a selfcontained presentation. Section 4 gives the final expressions for the polarization of the scattered neutrons, discusses special sector averages and shows the results for the 2π azimuthally averaged saturated state. Section 5 furnishes the details of the polarized SANS experiment and on the investigated sample, while Section 6 presents and discusses the analysis of the experimental results. Section 7 summarizes the main findings of this study. Appendix A features the expressions for the spin-resolved SANS cross sections in terms of the Fourier components of the magnetization, which enter the final expressions for the polarization, while Appendix B showcases some computed examples for the polarization.

2. Uniaxial SANS polarization analysis

Fig. 1 depicts a typical uniaxial neutron polarization analysis setup. We consider the most relevant cases where the externally applied magnetic field \mathbf{H}_0 , which defines the polarization axis for both the incident and scattered neutrons, is either perpendicular or parallel to the wavevector \mathbf{k}_0 of the incident neutron beam. Note that in both scattering geometries \mathbf{H}_0 is assumed to be parallel to the \mathbf{e}_z direction of a Cartesian laboratory coordinate system.

In a classical picture, the polarization **P** of a neutron beam containing N spins can be defined as the average over the individual polarizations \mathbf{P}_i of the neutrons (Schweizer, 2006):

$$\mathbf{P} = \frac{1}{N} \sum_{j=1}^{N} \mathbf{P}_j,\tag{1}$$

where $0 \le |\mathbf{P}| \le 1$. In experimental SANS studies the beam is usually partially polarized along a certain guide-field direction (quantization axis), which we take here as the *z* direction. Assuming that the expectation values of the perpendicular polarization components vanish, *i.e.* $P_x = P_y = 0$, and that $P_z = P$, one can then introduce the fractions

$$p^+ = \frac{1}{2}(1+P)$$
 and $p^- = \frac{1}{2}(1-P)$ (2)

of neutrons in the spin-up (+) and spin-down (-) states, with

$$p^+ + p^- = 1$$
 and $p^+ - p^- = P$. (3)

Obviously, for an unpolarized beam $p^+ = p^- = 0.5$ and P = 0, while P = +1 ($p^+ = 1$) or P = -1 ($p^- = 1$) for a fully polarized beam.

When there is an additional analyzer behind the sample, configured such that it selects only neutrons with spins either parallel or antiparallel to the initial polarization, then one can distinguish four scattering cross sections (scattering processes) (Blume, 1963; Moon *et al.*, 1969; Schweizer, 2006): two that conserve the neutron-spin direction (++ and --), called the non-spin-flip cross sections

$$\frac{1\Sigma^{++}}{\mathrm{d}\Omega} = K \Big[b_{\mathrm{H}}^{-2} |\widetilde{N}|^2 + b_{\mathrm{H}}^{-1} (\widetilde{N} \widetilde{Q}_z^* + \widetilde{N}^* \widetilde{Q}_z) + |\widetilde{Q}_z|^2 \Big],$$

$$\frac{1\Sigma^{--}}{\mathrm{d}\Omega} = K \Big[b_{\mathrm{H}}^{-2} |\widetilde{N}|^2 - b_{\mathrm{H}}^{-1} (\widetilde{N} \widetilde{Q}_z^* + \widetilde{N}^* \widetilde{Q}_z) + |\widetilde{Q}_z|^2 \Big], \quad (4)$$

and two cross sections which reverse the neutron spin (+- and -+), called the spin-flip cross sections

$$\frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega} = K \Big[|\widetilde{Q}_x|^2 + |\widetilde{Q}_y|^2 - i(\widetilde{Q}_x \widetilde{Q}_y^* - \widetilde{Q}_x^* \widetilde{Q}_y) \Big], \\ \frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega} = K \Big[|\widetilde{Q}_x|^2 + |\widetilde{Q}_y|^2 + i(\widetilde{Q}_x \widetilde{Q}_y^* - \widetilde{Q}_x^* \widetilde{Q}_y) \Big].$$
(5)



Figure 1

Sketch of the SANS setup and of the two most often employed scattering geometries in magnetic SANS experiments. (a) Applied magnetic field \mathbf{H}_{0} perpendicular to the incident neutron beam $(\mathbf{k}_0 \perp \mathbf{H}_0)$; (b) $\mathbf{k}_0 \parallel \mathbf{H}_0$. The momentum-transfer or scattering vector q corresponds to the difference between the wavevectors of the incident (\mathbf{k}_0) and the scattered (\mathbf{k}_1) neutrons, *i.e.* $\mathbf{q} = \mathbf{k}_0 - \mathbf{k}_1$. Its magnitude for elastic scattering, $q = |\mathbf{q}| = (4\pi/\lambda)\sin(\psi)$, depends on the mean wavelength λ of the neutrons and on the scattering angle 2ψ . For a given λ , sample-todetector distance $L_{\rm SD}$ and distance $r_{\rm D}$ from the centre of the direct beam to a certain pixel element on the detector, the q value can be obtained using $q \cong k_0(r_{\rm D}/L_{\rm SD})$. The symbols 'P', 'F' and 'A' denote, respectively, the polarizer, spin flipper and analyzer, which are optional neutron optical devices. Note that a second flipper after the sample has been omitted here. In spin-resolved SANS (POLARIS) using a ³He spin filter, the transmission (polarization) direction of the analyzer can be switched by 180° by means of a radiofrequency pulse. SANS is usually implemented as elastic scattering $(k_0 = k_1 = 2\pi/\lambda)$, and the component of **q** along the incident neutron beam [*i.e.* $q_x = 0$ in (a) and $q_z = 0$ in (b)] is neglected. The angle θ may be conveniently used in order to describe the angular anisotropy of the recorded scattering pattern on a twodimensional position-sensitive detector. Image taken from Michels (2021), reproduced by permission of Oxford University Press.

In these expressions, $K = 8\pi^3 V^{-1} b_{\rm H}^2$, where V denotes the scattering volume. $b_{\rm H} = 2.70 \times 10^{-15} \text{ m } \mu_{\rm B}^{-1} = 2.91 \times 10^8 \text{ A}^{-1} \text{ m}^{-1}$ is a constant (with $\mu_{\rm B}$ the Bohr magneton) that relates the atomic magnetic moment $\mu_{\rm a}$ to the atomic magnetic scattering length $b_{\rm m}$, given by (Moon *et al.*, 1969)

$$b_{\rm m} = \frac{\gamma_{\rm n} r_0}{2} \frac{\mu_{\rm a}}{\mu_{\rm B}} f(\mathbf{q}) \cong 2.70 \times 10^{-15} \,{\rm m} \frac{\mu_{\rm a}}{\mu_{\rm B}} f(\mathbf{q}) \cong b_{\rm H} \mu_{\rm a}.$$
 (6)

 $\gamma_n = 1.913$ denotes the neutron magnetic moment expressed in units of the nuclear magneton, $r_0 = 2.818 \times 10^{-15}$ m is the classical radius of the electron and $f(\mathbf{q})$ is the normalized atomic magnetic form factor, which we set to unity, $f \approx 1$, along the forward direction. The function $\tilde{N}(\mathbf{q})$ denotes the Fourier transform of the nuclear scattering-length density $N(\mathbf{r})$. The partial SANS cross sections, equations (4) and (5), are written here in terms of the Cartesian components of the Halpern–Johnson vector $\tilde{\mathbf{Q}}$ (sometimes also denoted as the magnetic interaction or magnetic scattering vector) (Halpern & Johnson, 1939):

$$\widetilde{\mathbf{Q}} = \widehat{\mathbf{q}} \times \left(\widehat{\mathbf{q}} \times \widetilde{\mathbf{M}} \right) = \widehat{\mathbf{q}} \left(\widehat{\mathbf{q}} \cdot \widetilde{\mathbf{M}} \right) - \widetilde{\mathbf{M}},\tag{7}$$

where $\hat{\mathbf{q}}$ is the unit scattering vector, and $\mathbf{M}(\mathbf{q}) = \{\widetilde{M}_x(\mathbf{q}), \widetilde{M}_y(\mathbf{q}), \widetilde{M}_z(\mathbf{q})\}$ represents the Fourier transform of the magnetization vector field $\mathbf{M}(\mathbf{r}) = \{M_x(\mathbf{r}), M_y(\mathbf{r}), M_z(\mathbf{r})\}$ of the sample under study. The three-dimensional Fourier-transform pair of the magnetization is defined as follows:

$$\widetilde{\mathbf{M}}(\mathbf{q}) = \frac{1}{(2\pi)^{3/2}} \int_{-\infty}^{+\infty} \int_{-\infty}^{+\infty} \int_{-\infty}^{+\infty} \mathbf{M}(\mathbf{r}) \exp(-i\mathbf{q} \cdot \mathbf{r}) \,\mathrm{d}^3 r, \quad (8)$$

$$\mathbf{M}(\mathbf{r}) = \frac{1}{(2\pi)^{3/2}} \int_{-\infty}^{+\infty} \int_{-\infty}^{+\infty} \int_{-\infty}^{+\infty} \widetilde{\mathbf{M}}(\mathbf{q}) \exp(i\mathbf{q} \cdot \mathbf{r}) \,\mathrm{d}^3 q. \quad (9)$$

The Halpern–Johnson vector is a manifestation of the dipolar origin of magnetic neutron scattering, and it emphasizes the fact that only the components of **M** that are perpendicular to **q** are relevant for magnetic neutron scattering. We note that different symbols for the Halpern–Johnson vector such as \mathbf{M}_{\perp} , \mathbf{Q}_{\perp} , \mathbf{S}_{\perp} or **q**, as in the original paper by Halpern & Johnson (1939), can be found in the literature. Likewise, in many textbooks (*e.g.* Lovesey, 1984; Squires, 2012) $\widetilde{\mathbf{Q}}$ is defined with a minus sign and normalized by the factor $2\mu_{\rm B}$, which makes it dimensionless. $\widetilde{\mathbf{Q}}$ is a linear vector function of the components of $\widetilde{\mathbf{M}}$. Both $\widetilde{\mathbf{Q}}(\mathbf{q})$ and $\widetilde{\mathbf{M}}(\mathbf{q})$ are in general complex vectors. For $\mathbf{k}_0 \perp \mathbf{H}_0$ and $\mathbf{k}_0 \parallel \mathbf{H}_0$ one finds, respectively (subscripts \perp and \parallel refer to the respective scattering geometry, compare Fig. 1),

$$\hat{\mathbf{q}}_{\perp} = \{0, \sin\theta, \cos\theta\},\tag{10}$$

$$\hat{\mathbf{q}}_{\parallel} = \{\cos\theta, \sin\theta, 0\}. \tag{11}$$

Inserting these expressions into equation (7) yields

$$\widetilde{\mathbf{Q}}_{\perp} = \begin{cases} -\widetilde{M}_{z} \\ -\widetilde{M}_{y}\cos^{2}\theta + \widetilde{M}_{z}\sin\theta\cos\theta \\ \widetilde{M}_{y}\sin\theta\cos\theta - \widetilde{M}_{z}\sin^{2}\theta \end{cases},$$
(12)

$$\widetilde{\mathbf{Q}}_{\parallel} = \begin{cases} -\widetilde{M}_{x}\sin^{2}\theta + \widetilde{M}_{y}\sin\theta\cos\theta\\ \widetilde{M}_{x}\sin\theta\cos\theta - \widetilde{M}_{y}\cos^{2}\theta\\ -\widetilde{M}_{z} \end{cases} \end{cases}.$$
 (13)

Inspection of equations (4) and (5) shows that the transverse components \tilde{Q}_x and \tilde{Q}_y give rise to spin-flip scattering, while the longitudinal component \tilde{Q}_z results in non-spin-flip scattering. Furthermore, if we set $\theta = 0^\circ$ in equation (12), which corresponds to the case that the scattering vector is along the neutron polarization, we see that

$$\widetilde{\mathbf{Q}}_{\perp}^{\theta=0^{\circ}} = \begin{cases} -\widetilde{M}_{x} \\ -\widetilde{M}_{y} \\ 0 \end{cases}, \qquad (14)$$

so that the magnetic scattering along the polarization direction is purely spin flip, and nuclear coherent and magnetic scattering are fully separated in the perpendicular scattering geometry [compare also with equations (4) and (5) for the non-spin-flip and the spin-flip SANS cross sections and with equation (21) for the final polarization]. In the case $\mathbf{k}_0 \parallel \mathbf{H}_0$ [equation (13)], spin-flip scattering probes only the transverse magnetization Fourier components $\widetilde{M}_{x,y}$, whereas the longitudinal scattering is entirely contained in the non-spin-flip channel, in contrast to the $\mathbf{k}_0 \perp \mathbf{H}_0$ geometry. We emphasize that nuclear-spin-dependent SANS is not taken into account in this paper, so that the corresponding scattering contributions do not show up in equations (4) and (5).

The total SANS cross section $d\Sigma/d\Omega$ can be expressed in terms of the initial spin populations p^{\pm} as (Blume, 1963; Moon *et al.*, 1969; Schweizer, 2006)

$$\frac{\mathrm{d}\Sigma}{\mathrm{d}\Omega} = p^+ \frac{\mathrm{d}\Sigma^{++}}{\mathrm{d}\Omega} + p^+ \frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega} + p^- \frac{\mathrm{d}\Sigma^{--}}{\mathrm{d}\Omega} + p^- \frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega}.$$
 (15)

Inserting the above expressions for p^+ and p^- [equations (2) and (3)] and for the partial SANS cross sections $d\Sigma^{\pm\pm}/d\Omega$ and $d\Sigma^{\pm\mp}/d\Omega$ [equations (4) and (5)], equation (15) evaluates to

$$\frac{\mathrm{d}\Sigma}{\mathrm{d}\Omega} = K \Big[b_{\mathrm{H}}^{-2} |\widetilde{N}|^2 + |\widetilde{\mathbf{Q}}|^2 + \mathbf{P} \cdot b_{\mathrm{H}}^{-1} (\widetilde{N} \widetilde{\mathbf{Q}}^* + \widetilde{N}^* \widetilde{\mathbf{Q}}) - i \mathbf{P} \cdot (\widetilde{\mathbf{Q}} \times \widetilde{\mathbf{Q}}^*) \Big],$$
(16)

which, using $\mathbf{P} = \{0, 0, P_z = \pm P\}$, can be rewritten as

$$\frac{\mathrm{d}\Sigma^{\pm}}{\mathrm{d}\Omega} = K \Big[b_{\mathrm{H}}^{-2} |\widetilde{N}|^{2} + |\widetilde{\mathbf{Q}}|^{2} \pm P b_{\mathrm{H}}^{-1} (\widetilde{N} \widetilde{Q}_{z}^{*} + \widetilde{N}^{*} \widetilde{Q}_{z}) \\ \mp i P (\widetilde{Q}_{x} \widetilde{Q}_{y}^{*} - \widetilde{Q}_{x}^{*} \widetilde{Q}_{y}) \Big].$$
(17)

Since the cross section is a scalar quantity and the polarization is an axial vector (or pseudovector), equation (16) shows that the system under study must itself contain an axial vector. As emphasized by Maleev (2002), examples for such built-in pseudovectors are related to the interaction of a polycrystalline sample with an external magnetic field (inducing an average magnetization directed along the applied field), the existence of a spontaneous magnetization in a ferromagnetic single crystal, the antisymmetric Dzyaloshinskii–Moriya interaction (DMI), mechanical (torsional) deformation or the presence of spin spirals. If on the other hand there is no preferred axis in the system, then $d\Sigma/d\Omega$ is independent of **P**. Examples include a collection of randomly oriented noninteracting nuclear (electronic) spins, which describe the general case of nuclear (paramagnetic) scattering at not-toolow temperatures and large applied fields, or a multi-domain ferromagnet with a random distribution of the domains. The same condition – existence of an axial system vector – applies for neutrons to be polarized in the scattering process [compare the last two terms in equation (21) below].

In the domain of magnetic SANS it is customary to denote experiments with a polarized incident beam only, and no spin analysis of the scattered neutrons, with the acronym SANSPOL. The two SANSPOL cross sections $d\Sigma^+/d\Omega$ and $d\Sigma^-/d\Omega$ (also sometimes denoted as the half-polarized SANS cross sections) combine non-spin-flip and spin-flip scattering contributions, according to $(p^{\pm} = 1)$

$$\frac{\mathrm{d}\Sigma^{+}}{\mathrm{d}\Omega} = \frac{\mathrm{d}\Sigma^{++}}{\mathrm{d}\Omega} + \frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega},\tag{18}$$

$$\frac{\mathrm{d}\Sigma^{-}}{\mathrm{d}\Omega} = \frac{\mathrm{d}\Sigma^{--}}{\mathrm{d}\Omega} + \frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega}.$$
(19)

Finally, noting that an unpolarized beam can be viewed as consisting of 50% spin-up and 50% spin-down neutrons [compare equations (2) and (3)], the unpolarized SANS cross section is obtained as [compare equation (15)]

$$\frac{\mathrm{d}\Sigma}{\mathrm{d}\Omega} = \frac{1}{2} \left(\frac{\mathrm{d}\Sigma^{++}}{\mathrm{d}\Omega} + \frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega} + \frac{\mathrm{d}\Sigma^{--}}{\mathrm{d}\Omega} + \frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega} \right)$$
$$= \frac{1}{2} \left(\frac{\mathrm{d}\Sigma^{+}}{\mathrm{d}\Omega} + \frac{\mathrm{d}\Sigma^{-}}{\mathrm{d}\Omega} \right).$$
(20)

The polarization $P_{\rm f}$ of the scattered beam along the direction of the incident neutron polarization P is obtained from the following relation (Blume, 1963; Moon *et al.*, 1969; Schweizer, 2006):

$$P_{\rm f} \frac{\mathrm{d}\Sigma}{\mathrm{d}\Omega} = p^+ \frac{\mathrm{d}\Sigma^{++}}{\mathrm{d}\Omega} + p^- \frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega} - p^- \frac{\mathrm{d}\Sigma^{--}}{\mathrm{d}\Omega} - p^+ \frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega}$$
$$= KP \Big(b_{\rm H}^{-2} |\widetilde{N}|^2 + |\widetilde{Q}_z|^2 - |\widetilde{Q}_x|^2 - |\widetilde{Q}_y|^2 \Big)$$
$$+ K \Big[b_{\rm H}^{-1} (\widetilde{N}\widetilde{Q}_z^* + \widetilde{N}^*\widetilde{Q}_z) + i(\widetilde{Q}_x\widetilde{Q}_y^* - \widetilde{Q}_x^*\widetilde{Q}_y) \Big].$$
(21)

The first four terms in the second line on the right-hand side of equation (21) demonstrate that nuclear scattering (to be more precise, the nuclear coherent scattering, the isotopic disorder scattering, and 1/3 of the nuclear-spin-dependent scattering) and scattering due to the longitudinal component \tilde{Q}_z of the magnetic scattering vector $\tilde{\mathbf{Q}}$ do not reverse the initial polarization, while the two transverse components \tilde{Q}_x and \tilde{Q}_y give rise to spin-flip scattering. The last two terms in equation (21) do create polarization: these are the familiar nuclear-magnetic interference term ($\tilde{N}\tilde{Q}_z^* + \tilde{N}^*\tilde{Q}_z$), which is commonly used to polarize beams, and the chiral term $i(\tilde{Q}_x\tilde{Q}_y^* - \tilde{Q}_x^*\tilde{Q}_y)$, which is of relevance in inelastic scattering (dynamic chirality) (Okorokov *et al.*, 1981; Maleev, 2002; Grigoriev *et al.*, 2015), in elastic scattering on spiral structures and weak ferromagnets (canted antiferromagnets) (Thoma *et al.*, 2021), or in the

presence of the DMI in microstructural-defect-rich magnets (Michels *et al.*, 2019; Quan *et al.*, 2020). We remind the reader that nuclear-spin-dependent scattering is not taken into account in the expressions for the magnetic SANS cross sections. In the general expression for the polarization of the scattered neutrons, a term $i\mathbf{P} \times (\widetilde{N}\mathbf{Q}^* - \widetilde{N}^*\mathbf{Q})$ appears (Schweizer, 2006), which is ignored in equation (21). This term rotates the polarization perpendicular to the initial polarization and cannot be observed in the uniaxial setup. We emphasize that in linear neutron polarimetry it is not possible to distinguish between a rotation of the polarization vector and a change of its length (Moon *et al.*, 1969; Maleev, 2002).

From equation (21) it follows that the polarization $P_{\rm f}(\mathbf{q})$ of the scattered neutron beam at momentum-transfer vector \mathbf{q} can be expressed as (Maleev *et al.*, 1963; Blume, 1963; Brown, 2006)

$$P_{\rm f} = \frac{p^+ \frac{\mathrm{d}\Sigma^{++}}{\mathrm{d}\Omega} + p^- \frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega} - p^- \frac{\mathrm{d}\Sigma^{--}}{\mathrm{d}\Omega} - p^+ \frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega}}{p^+ \frac{\mathrm{d}\Sigma^{++}}{\mathrm{d}\Omega} + p^+ \frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega} + p^- \frac{\mathrm{d}\Sigma^{--}}{\mathrm{d}\Omega} + p^- \frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega}}, \quad (22)$$

which for $p^+ = 1$ ($p^- = 0$) and $p^- = 1$ ($p^+ = 0$) reduces to, respectively,

$$P_{\rm f}^{+} = \frac{\frac{\mathrm{d}\Sigma^{++}}{\mathrm{d}\Omega} - \frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega}}{\frac{\mathrm{d}\Sigma^{++}}{\mathrm{d}\Omega} + \frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega}} = 1 - 2\frac{\frac{\mathrm{d}\Sigma^{+-}}{\mathrm{d}\Omega}}{\frac{\mathrm{d}\Sigma^{+}}{\mathrm{d}\Omega}}, \qquad (23a)$$

$$P_{\rm f}^{-} = \frac{\frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega} - \frac{\mathrm{d}\Sigma^{--}}{\mathrm{d}\Omega}}{\frac{\mathrm{d}\Sigma^{--}}{\mathrm{d}\Omega} + \frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega}} = -\left(1 - 2\frac{\frac{\mathrm{d}\Sigma^{-+}}{\mathrm{d}\Omega}}{\frac{\mathrm{d}\Sigma^{-}}{\mathrm{d}\Omega}}\right). \tag{23b}$$

Note that, for the following analysis, we drop the minus sign in front of the round brackets in equation (23*b*). For a quantitative analysis of $P_{\rm f}^{\pm}$, a theoretical model for the magnetization Fourier components $\widetilde{M}_{x,y,z}(\mathbf{q})$ and for $\widetilde{N}(\mathbf{q})$ is required. This will be discussed in the next section.

3. Sketch of micromagnetic SANS theory

Michels *et al.* (2016) presented a theory for the magnetic SANS cross section of bulk ferromagnets. The approach, which considers the response of the magnetization to spatially inhomogeneous magnetic anisotropy fields and magnetostatic fields, is based on the continuum theory of micromagnetics, valid close to magnetic saturation, and takes the antisymmetric DMI into account. Here, we recall the basic steps and ideas. The starting point is the static equations of micromagnetics for the bulk, which can be derived from the following expression for the magnetic Gibbs free energy (Brown, 1963; Aharoni, 2000; Kronmüller & Fähnle, 2003):

$$G = \int \boldsymbol{\epsilon} \, \mathrm{d}V$$

= $\int [A(\nabla \mathbf{m})^2 + D \, \mathbf{m} \cdot (\nabla \times \mathbf{m}) + \omega_{\mathrm{a}} - \frac{1}{2} \mu_0 \mathbf{H}_{\mathrm{d}} \cdot \mathbf{M} - \mu_0 \mathbf{H}_0 \cdot \mathbf{M}] \, \mathrm{d}V,$ (24)

where the first term denotes the energy due to the isotropic exchange interaction, the second term is the antisymmetric Dzyaloshinskii–Moriya energy (assuming a cubic symmetry), ω_a is the anisotropy energy density, and the last two terms are the energies related, respectively, to the dipolar interaction and the externally applied magnetic field; $\mathbf{m}(\mathbf{r}) = \mathbf{M}(\mathbf{r})/M_s(\mathbf{r})$ (Michels *et al.*, 2016). Variational calculus then leads to the following balance-of-torques equation:

$$\mathbf{M}(\mathbf{r}) \times \mathbf{H}_{\rm eff}(\mathbf{r}) = 0, \tag{25}$$

which expresses the fact that at static equilibrium the torque on the magnetization $\mathbf{M}(\mathbf{r})$ due to an effective magnetic field $\mathbf{H}_{\rm eff}(\mathbf{r})$ vanishes everywhere inside the material. The effective field is defined as the functional derivative of the ferromagnet's total energy-density functional ϵ with respect to the magnetization:

$$\begin{aligned} \mathbf{H}_{\rm eff} &= -\frac{1}{\mu_0} \frac{\delta \epsilon}{\delta \mathbf{M}} \\ &= \mathbf{H}_{\rm ex} + \mathbf{H}_{\rm DMI} + \mathbf{H}_{\rm p} + \mathbf{H}_{\rm d} + \mathbf{H}_0, \end{aligned} \tag{26}$$

where $\mathbf{H}_{ex} = l_M^2 \nabla^2 \mathbf{M}$ is the exchange field, $\mathbf{H}_{DMI} = -l_D \nabla \times \mathbf{M}$ is due to the DMI, $\mathbf{H}_p(\mathbf{r})$ is the magnetic anisotropy field, $\mathbf{H}_d(\mathbf{r})$ is the magnetostatic field and \mathbf{H}_0 is a uniform applied magnetic field; ∇^2 is the Laplace operator and $\nabla = \partial/\partial x \, \mathbf{e}_x + \partial/\partial y \, \mathbf{e}_y + \partial/\partial z \, \mathbf{e}_z$ is the gradient operator, where the unit vectors along the Cartesian laboratory axes are, respectively, denoted by \mathbf{e}_x , \mathbf{e}_y and \mathbf{e}_z (μ_0 , vacuum permeability). The micromagnetic length scales

$$l_{\rm M} = \left(\frac{2A}{\mu_0 M_0^2}\right)^{1/2} \tag{27}$$

and

$$l_{\rm D} = \frac{2D}{\mu_0 M_0^2}$$
(28)

are, respectively, related to the magnetostatic interaction and to the DMI. The values for the DMI constant D and for the exchange-stiffness constant A are assumed to be uniform throughout the material, in contrast to the local saturation magnetization $M_{\rm s}(\mathbf{r})$, which is assumed to depend explicitly on the position \mathbf{r} [see also Metlov & Michels (2015)]; $M_0 = V^{-1} \int_V M_{\rm s}(\mathbf{r}) \, dV$ denotes the macroscopic saturation magnetization of the sample, which can be measured with a magnetometer. Typical values for the above length scales are $l_{\rm M} \cong 5$ -10 nm (Kronmüller & Fähnle, 2003) and $l_{\rm D} \cong 1$ -2 nm. However, due to the lack of an established database for Dvalues, this estimate should be considered with some care.

The solution of the linearized version of equation (25) [see Michels *et al.* (2016) for details] provides closed-form expressions for the transverse Fourier components $\widetilde{M}_x(\mathbf{q})$ and $\widetilde{M}_y(\mathbf{q})$. Together with theoretical models (or even experimental data) for the longitudinal magnetization Fourier component $\widetilde{M}_z(\mathbf{q})$ and for the nuclear scattering amplitude $\widetilde{N}(\mathbf{q})$, these analytical solutions can be employed to compute the corresponding terms in the SANS cross sections (see Appendix A) and, hence, to evaluate the final polarizations.

Averaging over the directions of the magnetic anisotropy field in the plane perpendicular to the applied field, the magnetic terms for the transverse magnetic field geometry ($\mathbf{k}_0 \perp \mathbf{H}_0$, $q_x = 0$) are

$$\begin{split} |M_{x}|^{2} &= \\ \frac{p^{2}}{2} \frac{\widetilde{H}_{p}^{2} \Big[\big(1 + p \sin^{2} \theta \big)^{2} + p^{2} l_{D}^{2} q^{2} \cos^{2} \theta \Big] + 2 \widetilde{M}_{z}^{2} (1 + p)^{2} l_{D}^{2} q^{2} \sin^{2} \theta}{\big(1 + p \sin^{2} \theta - p^{2} l_{D}^{2} q^{2} \cos^{2} \theta \big)^{2}}, \end{split}$$

$$(29)$$

$$\begin{split} |M_{y}|^{2} &= \\ \frac{p^{2}}{2} \frac{\widetilde{H}_{p}^{2}(1+p^{2}l_{D}^{2}q^{2}\cos^{2}\theta) + 2\widetilde{M}_{z}^{2}(1+pl_{D}^{2}q^{2})^{2}\sin^{2}\theta\cos^{2}\theta}{\left(1+p\sin^{2}\theta-p^{2}l_{D}^{2}q^{2}\cos^{2}\theta\right)^{2}}, \quad (30) \end{split}$$

$$CT_{yz} = \widetilde{M}_{y}\widetilde{M}_{z}^{*} + \widetilde{M}_{y}^{*}\widetilde{M}_{z} = -\frac{2\widetilde{M}_{z}^{2}p(1+pl_{D}^{2}q^{2})\sin\theta\cos\theta}{1+p\sin^{2}\theta-p^{2}l_{D}^{2}q^{2}\cos^{2}\theta},$$
(31)

$$-2i\chi = \frac{2\widetilde{H}_{p}^{2}p^{3}(2+p\sin^{2}\theta)l_{D}q\cos^{3}\theta + 4\widetilde{M}_{z}^{2}p(1+p)^{2}l_{D}q\sin^{2}\theta\cos\theta}{\left(1+p\sin^{2}\theta-p^{2}l_{D}^{2}q^{2}\cos^{2}\theta\right)^{2}}.$$
(32)

The results for the parallel field geometry $(\mathbf{k}_0 \parallel \mathbf{H}_0, q_z = 0)$ are

$$|\widetilde{M}_{x}|^{2} = \frac{p^{2}}{2} \frac{\widetilde{H}_{p}^{2} \left[1 + p(2+p)\sin^{2}\theta\right] + 2\widetilde{M}_{z}^{2}(1+p)^{2} l_{D}^{2} q^{2} \sin^{2}\theta}{(1+p)^{2}},$$
(33)

$$|\widetilde{M}_{y}|^{2} = \frac{p^{2}}{2} \frac{\widetilde{H}_{p}^{2} \left[1 + p(2+p)\cos^{2}\theta\right] + 2\widetilde{M}_{z}^{2}(1+p)^{2} l_{D}^{2} q^{2}\cos^{2}\theta}{(1+p)^{2}},$$
(34)

$$CT_{xy} = \widetilde{M}_{x}\widetilde{M}_{y}^{*} + \widetilde{M}_{x}^{*}\widetilde{M}_{y}$$
$$= -p^{2} \frac{\left[\widetilde{H}_{p}^{2}p(2+p) + 2\widetilde{M}_{z}^{2}(1+p)^{2}l_{D}^{2}q^{2}\right]\sin\theta\cos\theta}{\left(1+p\right)^{2}}, (35)$$

$$\chi = 0. \tag{36}$$

In equations (29)–(36), $\tilde{H}_p^2(q\xi_{\rm H})$ denotes the magnitudesquare of the Fourier transform of the magnetic anisotropy field, and $\tilde{M}_z^2(q\xi_{\rm M})$ is the Fourier component of the longitudinal magnetization. We also emphasize that

$$\widetilde{M}_z \cong \widetilde{M}_s$$
 (37)

is assumed in the approach-to-saturation regime, which Bersweiler *et al.* (2022) defined for applied fields where the reduced magnetization $M/M_0 \gtrsim 90\%$. These functions characterize the strength and spatial structure of, respectively, the magnetic anisotropy field $\mathbf{H}_{p}(\mathbf{r})$, with correlation length $\xi_{\rm H}$, and the local saturation magnetization $M_{\rm s}(\mathbf{r})$, with correlation length $\xi_{\rm M}$.

$$p(q, H_{\rm i}) = \frac{M_0}{H_{\rm eff}(q, H_{\rm i})} = \frac{M_0}{H_{\rm i}(1 + l_{\rm H}^2 q^2)}$$
(38)

is a dimensionless function, where

$$H_{\rm eff}(q, H_{\rm i}) = H_{\rm i} (1 + l_{\rm H}^2 q^2)$$
 (39)

is the effective magnetic field [not to be confused with $\mathbf{H}_{\text{eff}}(\mathbf{r})$ in equation (25)], which depends on the internal magnetic field $H_i = H_0 - NM_0$ (N, demagnetizing factor), on $q = |\mathbf{q}|$ and on the exchange length of the field

$$l_{\rm H} = \left(\frac{2A}{\mu_0 M_0 H_{\rm i}}\right)^{1/2}.$$
 (40)

The latter quantity is a measure for the size of inhomogeneously magnetized regions around microstructural lattice defects (Mettus & Michels, 2015). The Fourier coefficient of the longitudinal magnetization $\widetilde{M}_{z}(\mathbf{q})$ provides information on the spatial variation of the saturation magnetization $M_{s}(\mathbf{r})$; for instance, $|\widetilde{M}_z|^2 \propto (\Delta M)^2$ in a multiphase magnetic nanocomposite, where ΔM denotes the jump of the magnetization magnitude at internal (particle-matrix) interfaces (Honecker & Michels, 2013). Moreover, while the squared Fourier components and the cross terms are even functions of q, it is easily seen that the chiral term $-2i\chi(\mathbf{q})$ [equation (32)] is asymmetric in q, which is due to the DMI term: at small fields, when the term $\propto \widetilde{H}_{p}^{2}$ in the numerator of equation (32) dominates, two extrema parallel and antiparallel to the field axis are observed, whereas at larger fields, when the term $\propto \tilde{M}_{z}^{2}$ dominates, additional maxima and minima appear approximately along the detector diagonals (Michels et al., 2016).

By inserting equations (29)-(36) into the SANS cross sections (see Appendix A) and by specifying particular models for the nuclear scattering function $\widetilde{N}^2(q\xi_{nuc})$, the longitudinal magnetic Fourier component $\tilde{M}_z^2(q\xi_M)$ and the Fourier coefficient of the magnetic anisotropy field $\widetilde{H}_{p}^{2}(q\xi_{H})$, one obtains P^{\pm} as a function of the magnitude and orientation of the scattering vector \mathbf{q} , the applied magnetic field \mathbf{H}_0 , the magnetic interaction parameters $(A, D, M_0, \Delta M, H_p, \xi_M, \xi_H)$ and microstructural quantities (particle-size distribution, crystallographic texture etc.). We emphasize that the expressions (29)-(36) for the Fourier components can of course also be employed directly in the SANS cross sections to analyze experimental scattering data [see e.g. Bersweiler et al. (2022) for a recent example]. Later on in the paper, for graphically displaying P^{\pm} , we have assumed that \widetilde{N}^2 , \widetilde{M}_2^2 and \widetilde{H}_p^2 are all isotropic (i.e. θ independent), as is appropriate for polycrystalline texture-free bulk ferromagnets, and that they can be represented by Lorentzian-squared functions, i.e.

$$\widetilde{M}_{z}^{2}(q\xi_{\rm M}) = \frac{A_{\rm M}^{2}\xi_{\rm M}^{6}}{\left(1 + q^{2}\xi_{\rm M}^{2}\right)^{2}},\tag{41}$$

$$\widetilde{H}_{\rm p}^2(q\xi_{\rm H}) = \frac{A_{\rm H}^2 \xi_{\rm H}^6}{\left(1 + q^2 \xi_{\rm H}^2\right)^2},\tag{42}$$

$$\widetilde{N}^2(q\xi_{\rm nuc}) = \alpha(q)b_{\rm H}^2 \widetilde{M}_z^2(q\xi_{\rm M}), \qquad (43)$$

where the amplitudes $A_{\rm M}^2$ and $A_{\rm H}^2$ (both in units of A² nm⁻²) are, respectively, related to the mean-square magnetization fluctuation and anisotropy-field variation. Of course, other scattering functions such as the form factor of a sphere and various structure-factor models (e.g. a Percus-Yevick hardsphere structure factor) can be straightforwardly implemented (Mettus & Michels, 2015). The characteristic structure sizes of \widetilde{M}_{z}^{2} and \widetilde{H}_{p}^{2} are generally different. We remind the reader that these are related, respectively, to the spatial extent of regions with uniform saturation magnetization ($\xi_{\rm M}$) and magnetic anisotropy field ($\xi_{\rm H}$). Measurement of the magnetic-fielddependent Guinier radius provides a means to determine these correlation lengths as well as the exchange-stiffness constant A (Michels et al., 2020). A simple example where $\xi_{\rm M} = \xi_{\rm H}$ is a collection of homogeneous and defect-free magnetic nanoparticles in a magnetic and homogeneous matrix. If, on the other hand, atomically sharp grain boundaries are introduced into such particles, then the direction of the magnetic anisotropy field changes due to the changing set of crystallographic directions at the intraparticle interfaces, but the value of M_s may remain the same, so that $\xi_{\rm H} < \xi_{\rm M}$. Honecker & Michels (2013) showed, assuming $\xi_{\rm H} = \xi_{\rm M}$ and using the sphere form factor for both \tilde{M}_z^2 and \tilde{H}_p^2 , that it is the ratio of $H_{\rm p}/\Delta M$ (related to the amplitudes $A_{\rm H}$ and $A_{\rm M}$) that determines the angular anisotropy and the asymptotic powerlaw dependence of $d\Sigma/d\Omega$ as well as the characteristic decay length of spin-misalignment fluctuations. The ratio of nuclear to longitudinal magnetic scattering is denoted by α , which for the general case (that the nuclear correlation length ξ_{nuc} is different from $\xi_{\rm M}$) is a function of q. Here, we do not specify a particular ξ_{nuc} and assume this characteristic size to be contained in $\alpha(q)$.

4. Polarization of scattered beam

When the expressions for the elastic differential spin-flip and SANSPOL cross sections $d\Sigma^{\pm\mp}/d\Omega$ and $d\Sigma^{\pm}/d\Omega$ [equations (66)–(70)] are inserted into equations (23*a*) and (23*b*), and use is made of the expressions for the magnetization Fourier components [equations (29)–(36)], one obtains, respectively, for the transverse and longitudinal scattering geometry

$$P_{f\perp}^{+}(\mathbf{q}) = \frac{h_{1}(\mathbf{q}) - i\chi(\mathbf{q})}{1 - 2\frac{h_{2}(\mathbf{q}) - b_{\mathrm{H}}^{-1}\mathrm{CT}_{\widetilde{NM}_{z}}\sin^{2}\theta + b_{\mathrm{H}}^{-1}\mathrm{CT}_{\widetilde{NM}_{y}}\sin\theta\cos\theta - i\chi(\mathbf{q})},$$
(44*a*)

$$P_{f\perp}^{-}(\mathbf{q}) = \frac{h_{1}(\mathbf{q}) + i\chi(\mathbf{q})}{1 - 2\frac{h_{2}(\mathbf{q}) + b_{H}^{-1}CT_{\widetilde{NM}_{z}}\sin^{2}\theta - b_{H}^{-1}CT_{\widetilde{NM}_{y}}\sin\theta\cos\theta + i\chi(\mathbf{q})},$$
(44b)

$$P_{\mathrm{f}\parallel}^{+}(\mathbf{q}) = 1 - 2 \frac{g_{1}(\mathbf{q})}{g_{2}(\mathbf{q}) - b_{\mathrm{H}}^{-1} \mathrm{CT}_{\widetilde{NM}_{z}}}, \qquad (44c)$$

$$P_{\mathrm{f}\parallel}^{-}(\mathbf{q}) = 1 - 2 \frac{g_{1}(\mathbf{q})}{g_{2}(\mathbf{q}) + b_{\mathrm{H}}^{-1} \mathrm{CT}_{\widetilde{NM_{*}}}}.$$
 (44*d*)

The functions $h_1(\mathbf{q})$, $h_2(\mathbf{q})$, $g_1(\mathbf{q})$ and $g_2(\mathbf{q})$ are independent of the incident neutron beam polarization and are defined as

$$h_1(\mathbf{q}) = |\widetilde{\boldsymbol{M}}_x|^2 + |\widetilde{\boldsymbol{M}}_y|^2 \cos^4 \theta + |\widetilde{\boldsymbol{M}}_z|^2 \sin^2 \theta \cos^2 \theta - CT_{yz} \sin \theta \cos^3 \theta, \quad (45)$$

$$h_{2}(\mathbf{q}) = b_{\mathrm{H}}^{-2} |\widetilde{N}|^{2} + |\widetilde{M}_{x}|^{2} + |\widetilde{M}_{y}|^{2} \cos^{2} \theta + |\widetilde{M}_{z}|^{2} \sin^{2} \theta - \mathrm{CT}_{yz} \sin \theta \cos \theta, \qquad (46)$$

$$g_1(\mathbf{q}) = |\widetilde{M}_x|^2 \sin^2 \theta + |\widetilde{M}_y|^2 \cos^2 \theta - CT_{xy} \sin \theta \cos \theta, \quad (47)$$

$$g_{2}(\mathbf{q}) = b_{\mathrm{H}}^{-2} |\tilde{N}|^{2} + |\tilde{M}_{x}|^{2} \sin^{2}\theta + |\tilde{M}_{y}|^{2} \cos^{2}\theta + |\tilde{M}_{z}|^{2} - \mathrm{CT}_{xy} \sin\theta\cos\theta.$$
(48)

At complete magnetic saturation, when $\mathbf{M}(\mathbf{r}) = \{0, 0, M_s(\mathbf{r})\}$, these expressions reduce to

$$h_1^{\text{sat}}(\mathbf{q}) = |\widetilde{\boldsymbol{M}}_{\text{s}}|^2 \sin^2 \theta \cos^2 \theta, \qquad (49)$$

$$h_2^{\text{sat}}(\mathbf{q}) = b_{\text{H}}^{-2} |\widetilde{N}|^2 + |\widetilde{M}_{\text{s}}|^2 \sin^2 \theta, \qquad (50)$$

$$g_1^{\text{sat}}(\mathbf{q}) = 0, \tag{51}$$

$$g_2^{\text{sat}}(\mathbf{q}) = b_{\text{H}}^{-2} |\widetilde{N}|^2 + |\widetilde{M}_{\text{s}}|^2, \qquad (52)$$

where $\widetilde{M}_{s}(\mathbf{q})$ is the Fourier transform of $M_{s}(\mathbf{r})$. As can be seen from equations (44*a*) and (44*b*), the difference between P_{f}^{+} and P_{f}^{-} resides, for $\mathbf{k}_{0} \perp \mathbf{H}_{0}$, in the nuclear-magnetic interference terms $\propto \widetilde{N}\widetilde{M}_{z}$ and $\propto \widetilde{N}\widetilde{M}_{y}$, and in $\chi(\mathbf{q})$, while for $\mathbf{k}_{0} \parallel \mathbf{H}_{0}$ the two polarizations differ only by the term $\propto \widetilde{N}\widetilde{M}_{z}$ [equations (44*c*) and (44*d*)]. We also remind the reader that the Fourier coefficients in the above expressions are evaluated in the plane of the detector, which for the perpendicular scattering geometry corresponds to the plane $q_{x} = 0$ and for the parallel geometry to the plane $q_{z} = 0$ (compare Fig. 1).

The \tilde{NM}_y contribution to equations (44*a*) and (44*b*) requires special consideration. This term is expected to be negligible for a polycrystalline statistically isotropic ferromagnet with vanishing fluctuations of the saturation magnetization. This can be seen by scrutinizing the following expression for the \tilde{M}_y magnetization Fourier component in the perpendicular scattering geometry, corresponding to the plane $q_x = 0$ (Michels, 2021):

$$\widetilde{M}_{y} = \frac{p \left[\widetilde{H}_{py} - \widetilde{M}_{z}(q_{y}q_{z}/q^{2})(1 + pl_{D}^{2}q^{2}) - i\widetilde{H}_{px}pl_{D}q_{z} \right]}{1 + p(q_{y}^{2}/q^{2}) - p^{2}l_{D}^{2}q_{z}^{2}},$$
 (53)

where \widetilde{H}_{px} and \widetilde{H}_{py} denote the Cartesian components of the Fourier transform of the magnetic anisotropy field. If we assume that the nuclear scattering is isotropic and that \widetilde{H}_{px} and \widetilde{H}_{py} vary randomly in the plane perpendicular to the field, then

the corresponding averages over the direction of the anisotropy field vanish. The only remaining term in the $\widetilde{N}\widetilde{M}_y$ contribution is then $(q_y/q = \sin\theta; q_z/q = \cos\theta)$

$$CT_{\widetilde{NM}_{y}} = \widetilde{N}\widetilde{M}_{y}^{*} + \widetilde{N}^{*}\widetilde{M}_{y}$$
$$= -\frac{2\widetilde{N}\widetilde{M}_{z}p(1+pl_{\mathrm{D}}^{2}q^{2})\sin\theta\cos\theta}{1+p\sin^{2}\theta - p^{2}l_{\mathrm{D}}^{2}q^{2}\cos^{2}\theta}, \qquad (54)$$

where we have furthermore assumed that both $\widetilde{N}(q)$ and $\widetilde{M}_z(q)$ are real valued, as is done throughout this paper. Note that equation (54) still needs to be multiplied with the term $\sin \theta \cos \theta$ in order to obtain the corresponding contribution to $P_{f\perp}^{\pm}$ [compare equations (44*a*) and (44*b*)]. For homogeneous single-phase materials with $M_s = \text{constant}$, the $\widetilde{N}\widetilde{M}_y$ contribution is expected to be negligible, and we are not aware that this has been reported experimentally. However, for materials exhibiting strong spatial nanoscale variations in the saturation magnetization, *i.e.* $M_s = M_s(\mathbf{r})$, such as magnetic nanocomposites or porous ferromagnets, it should be possible to observe this scattering contribution in polarized SANS experiments.

4.1. Sector averages

Carrying out a 2π azimuthal average of $P_{\rm f}^{\pm}(\mathbf{q})$, which are maps with numbers varying between ± 1 , may result in a significant loss of information (compare *e.g.* Figs. 10 and 11 below). It is therefore often advantageous to consider cuts of $P_{\rm f}^{\pm}$ along certain directions in \mathbf{q} space. This might also be of relevance for other spin-manipulating techniques such as SEMSANS, which is a one-dimensional technique that only measures correlations in the encoding direction (Li *et al.*, 2021). However, one has to keep in mind that SEMSANS is a real-space technique that (similar to SESANS and DFI) essentially measures the cosine Fourier transform of the cross section. Nevertheless, the analytical expressions for the magnetization Fourier components can be used in such a transform to obtain information on the magnetic interactions via the projected correlation function.

Sector averages are straightforwardly obtained by evaluating equations (44*a*)–(44*d*) [using equations (45)–(52)] for certain angles θ . For instance, for the perpendicular scattering geometry and for **q** along the vertical direction on the detector ($\theta = 90^\circ$), we obtain [$\chi(q, \theta = 90^\circ) = 0$, compare equation (32)]

$$P_{f\perp}^{+}(q,\theta = 90^{\circ}) = 1 - 2 \frac{|\widetilde{M}_{x}|^{2}}{b_{\rm H}^{-2}|\widetilde{N}|^{2} + |\widetilde{M}_{x}|^{2} + |\widetilde{M}_{z}|^{2} - b_{\rm H}^{-1}\mathrm{CT}_{\widetilde{NM}_{z}}},$$
(55a)

$$P_{\mathrm{f}\perp}^{-}(q,\theta=90^{\circ}) = 1 - 2 \frac{|\widetilde{\boldsymbol{M}}_{x}|^{2}}{b_{\mathrm{H}}^{-2}|\widetilde{\boldsymbol{N}}|^{2} + |\widetilde{\boldsymbol{M}}_{x}|^{2} + |\widetilde{\boldsymbol{M}}_{z}|^{2} + b_{\mathrm{H}}^{-1}\mathrm{CT}_{\widetilde{\boldsymbol{N}}\widetilde{\boldsymbol{M}}_{z}}},$$
(55b)

where [compare equation (29)]

$$|\widetilde{M}_{x}|^{2}(q,\theta = 90^{\circ}) = \frac{p^{2}}{2} \left(\widetilde{H}_{p}^{2} + 2\widetilde{M}_{z}^{2} l_{D}^{2} q^{2} \right).$$
(56)

At saturation $(M_x = 0)$, $P_{f\perp}^{\pm}(q, \theta = 90^{\circ}) = 1$, except for the case $\alpha = 1$, where $P_{f\perp}^{+}(q, \theta = 90^{\circ}) = -1$. We also see that information on the DMI is contained in $|\widetilde{M}_x|^2$ via the length scale $l_{\rm D}$. For $l_{\rm D} = 0$, $|\widetilde{M}_x|^2 = (p^2/2)\widetilde{H}_{\rm p}^2$.

For the perpendicular scattering geometry and **q** along the horizontal direction ($\theta = 0^{\circ}$), we obtain

$$P_{f\perp}^{+}(q,\theta=0^{\circ}) = 1 - 2\frac{|\widetilde{M}_{x}|^{2} + |\widetilde{M}_{y}|^{2} - i\chi}{b_{\rm H}^{-2}|\widetilde{N}|^{2} + |\widetilde{M}_{x}|^{2} + |\widetilde{M}_{y}|^{2} - i\chi}, \quad (57a)$$

$$P_{\rm f\perp}^{-}(q,\theta=0^{\circ}) = 1 - 2 \frac{|\widetilde{M}_{x}|^{2} + |\widetilde{M}_{y}|^{2} + i\chi}{b_{\rm H}^{-2}|\widetilde{N}|^{2} + |\widetilde{M}_{x}|^{2} + |\widetilde{M}_{y}|^{2} + i\chi}, \quad (57b)$$

where [compare equations (29) and (30)]

$$|\widetilde{M}_{x}|^{2}(q,\theta=0^{\circ}) = |\widetilde{M}_{y}|^{2} = \frac{p^{2}}{2} \frac{\widetilde{H}_{p}^{2}(1+p^{2}l_{D}^{2}q^{2})}{\left(1-p^{2}l_{D}^{2}q^{2}\right)^{2}}$$
(58)

and [compare equation (32)]

$$i\chi(q,\theta=0^{\circ}) = -\frac{2\tilde{H}_{\rm p}^2 p^3 l_{\rm D} q}{\left(1-p^2 l_{\rm D}^2 q^2\right)^2}.$$
(59)

At saturation $(M_x = M_y = \chi = 0)$ and for nonzero nuclear scattering, $P_{f\perp}^{\pm}(q, \theta = 0^{\circ}) = 1$. For $l_{\rm D} = 0$, $|\widetilde{M}_x|^2 + |\widetilde{M}_y|^2 = p^2 \widetilde{H}_{\rm p}^2$ and $P_{f\perp}^{\pm}(q, \theta = 0^{\circ})$ contains information on the transverse spin components.

4.2. Saturated state

At saturation and for $\mathbf{k}_0 \perp \mathbf{H}_0$, it is readily verified from equations (44*a*) and (44*b*) using equations (49)–(52), $\chi(\mathbf{q}) = 0$ and $\operatorname{CT}_{\widetilde{NM}_v} = 0$ that

$$P_{f\perp}^+(q,\theta) = 1 - 2\frac{\sin^2\theta\cos^2\theta}{\alpha - 2\alpha^{1/2}\sin^2\theta + \sin^2\theta},\tag{60a}$$

$$P_{\rm f\perp}^{-}(q,\theta) = 1 - 2\frac{\sin^2\theta\cos^2\theta}{\alpha + 2\alpha^{1/2}\sin^2\theta + \sin^2\theta} \tag{60b}$$

depend exclusively on the ratio

$$\alpha(q) = \frac{\widetilde{N}^2(q)}{b_{\rm H}^2 \widetilde{M}_{\rm s}^2(q)} \tag{61}$$

of nuclear to longitudinal magnetic scattering [compare equation (43)]. The possible angular anisotropy of α is not considered in this paper. Since for $\mathbf{k}_0 \parallel \mathbf{H}_0$ the spin-flip SANS cross section vanishes at saturation $(g_1 = 0)$, we see that $P_{\text{fl}}^{\pm} = +1$. The azimuthally averaged $[(1/2\pi) \int (\cdots) d\theta]$ versions of equations (60*a*) and (60*b*) read

$$P_{f\perp}^{+}(q) = \frac{2\alpha^{1/2} \left(-1 + \alpha^{1/2} + \left|-1 + \alpha^{1/2}\right|\right)}{\left(1 - 2\alpha^{1/2}\right)^{2}},$$
 (62a)

$$P_{\rm f\perp}^{-}(q) = 1 - \frac{1}{\left(1 + 2\alpha^{1/2}\right)^2}.$$
 (62b)

The function $\alpha(q)$ can be a monotonically increasing or decreasing function of q, and it can even exhibit local extrema. In the following, we will consider the cases of $\alpha = \text{constant}$ and $\alpha = \alpha(q)$ using the experimental data of the soft magnetic Fe-based alloy NANOPERM (Michels *et al.*, 2012).

4.2.1. α = constant. Fig. 2 displays the two-dimensional polarization $P_{f\perp}^{\pm}(\mathbf{q})$ of the scattered neutrons in the saturated state as a function of α = constant. The case of constant α is very rarely realized in experimental situations, and we consider it here only as a starting point for our discussion and for the comparison with the experimentally more relevant situation of $\alpha = \alpha(q)$. For $\alpha \to 0$ it follows that $P_{f\perp}^{\pm} = 1 - 2\cos^2\theta$ [Figs. 2(α) and 2(e)], while $P_{f\perp}^{\pm} = 1 - 2\sin^2\theta$ [Fig. 2(c)] and $P_{f\perp}^{-} = 1 - 2\sin^2\theta \cos^2\theta/(1 + 3\sin^2\theta)$ [Fig. 2(g)] for $\alpha = 1$. When nuclear coherent scattering is dominating $(\alpha \to \infty)$, we see that $P_{f\perp}^{\pm}$ both tend to unity, as expected. The corresponding 2π azimuthally averaged functions [equations (62a) and (62b)] are plotted in Fig. 3. One readily verifies that $P_{f\perp}^{\pm} = 0$ for $\alpha^{1/2} \leq 1$, which further underlines the loss of



Plot of $P_{f_1}^+(q_v, q_z)$ (upper row) and $P_{f_1}^-(q_v, q_z)$ (lower row) in the saturated state for different values of α (see insets) [equations (60a) and (60b)].

Figure 2

information when a 2π azimuthal average is carried out [compare *e.g.* Fig. 2(*b*)].

4.2.2. a = a(q). Fig. 4 shows the experimentally determined ratio $\alpha_{exp}(q)$ (Michels *et al.*, 2012) of nuclear to magnetic



Figure 3

Plot of $P_{f\perp}^+$ and $P_{f\perp}^-$ (see inset) in the saturated state as a function of α [equations (62*a*) and (62*b*)].



Figure 4

Black circles: experimental ratio $\alpha_{\exp}(q)$ of nuclear to magnetic scattering of the two-phase alloy NANOPERM (Michels *et al.*, 2012) ($\mathbf{k}_0 \perp \mathbf{H}_0$; $\mu_0 H_0 = 1.27$ T; log–log plot). Solid line: power-law fit to parametrize the experimental data [equation (63)]. The fit has been restricted to the interval 0.03 < q < 0.3 nm⁻¹, but the fit function is displayed for 0.01 < q <1.0 nm⁻¹.



Figure 5

Plot of $P_{f\perp}^+(q)$ (solid line) and $P_{f\perp}^-(q)$ (dashed line) of NANOPERM using equations (62*a*) and (62*b*) with $\alpha_{\exp}(q)$ given by equation (63) and 0.03 < $q < 0.3 \text{ nm}^{-1}$.

scattering of the two-phase alloy NANOPERM. Within the experimental q range of $0.03 < q < 0.3 \text{ mm}^{-1}$, these data for $\alpha_{\exp}(q)$ have been fitted by a power law in 1/q to obtain the functions $P_{f\perp}^{\pm}(q)$, which are depicted in Fig. 5. The used fit function for $\alpha_{\exp}(q)$ is

$$\alpha_{\exp}(q) = \frac{0.14853}{q} - \frac{0.0264491}{q^2} + \frac{0.00176887}{q^3} - \frac{4.95094 \times 10^{-5}}{q^4} + \frac{5.01767 \times 10^{-7}}{q^5}.$$
 (63)

This expression will be used in the analysis of the experimental data (see Section 6).

4.3. Nonsaturated state

Appendix *B* features some theoretical results for $P_{f\perp}^{\pm}(\mathbf{q})$ for various combinations of the magnetic interaction parameters [applied magnetic field, ratio of \widetilde{H}_p^2 to \widetilde{M}_z^2 , $\chi(\mathbf{q}) \neq 0$]. For a statistically isotropic ferromagnet, the two-dimensional distribution of the polarization of the scattered neutrons is isotropic (θ independent) for the longitudinal scattering geometry ($\mathbf{k}_0 \parallel \mathbf{H}_0$), as are the corresponding SANS cross sections. This is in contrast to the $P_{f\perp}^{\pm}(\mathbf{q})$ for the transverse geometry ($\mathbf{k}_0 \perp \mathbf{H}_0$), which are highly anisotropic. In the following, we will use the theoretical expressions for $P_{f\perp}^{\pm}(\mathbf{q})$ to analyze experimental data on the soft magnetic two-phase nanocrystalline alloy NANOPERM.

5. Experimental details

The polarized neutron experiment was carried out at room temperature at the instrument D22 at the Institut Laue-Langevin, Grenoble, France. Incident neutrons with a mean wavelength of $\lambda = 8 \text{ Å}$ and a wavelength broadening of $\Delta\lambda/\lambda = 10\%$ (FWHM) were selected by means of a velocity selector. The beam was polarized using a 1.2 m-long remanent Fe–Si supermirror transmission polarizer (m = 3.6), which was installed immediately after the velocity selector. A radiofrequency spin flipper, installed close to the sample position, allowed us to reverse the initial neutron polarization. The external magnetic field (provided by an electromagnet) was applied perpendicular to the wavevector \mathbf{k}_0 of the incident neutrons (compare Fig. 1). Measurement of the four partial POLARIS cross sections $d\Sigma^{++}/d\Omega$, $d\Sigma^{--}/d\Omega$, $d\Sigma^{+-}/d\Omega$ and $d\Sigma^{-+}/d\Omega$ was accomplished through a polarized ³He spinfilter cell, which was installed inside the detector housing, about 1 m away from the sample position. The polarization between polarizer, radiofrequency flipper and ³He filter was maintained by means of magnetic guide fields of the order of 1 mT. The efficiencies of the polarizer, spin flipper and 3 He analyzer were, respectively, 90, 99 and 87.5%. The scattered neutrons were detected by a multitube detector which consists of 128×128 pixels with a resolution of 8×8 mm. Neutron data reduction, including corrections for background scattering and spin leakage (Wildes, 2006), was performed using the GRASP (Dewhurst, 2021) and BerSANS (Keiderling, 2002; Keiderling et al., 2008) software packages.

The sample under study was a two-phase magnetic nanocomposite from the NANOPERM family of alloys (Suzuki & Herzer, 2006) with a nominal composition of $(Fe_{0.985}Co_{0.015})_{90}$ -Zr₇B₃ (Suzuki *et al.*, 1994; Ito *et al.*, 2007). The alloy was prepared by melt spinning, followed by a subsequent annealing treatment for 1 h at 883 K, which resulted in the precipitation of body-centered cubic iron nanoparticles in an amorphous magnetic matrix. The average iron particle size of $D = 15 \pm 2$ nm was determined by the analysis of wide-angle X-ray diffraction data. The crystalline particle volume fraction is about 65% and the saturation magnetization of the alloy amounts to $\mu_0 M_0 = 1.64$ T. The exchange-stiffness constant $A = (4.7 \pm 0.9) \times 10^{-12} \text{ J m}^{-1}$ has previously been determined by the analysis of the field-dependent unpolarized SANS cross section (Honecker *et al.*, 2013). For the SANS experiments, several circular discs with a diameter of 10 mm and a thickness of about 20 µm were stacked and mounted on a Cd aperture [for further details see Michels *et al.* (2012) and Honecker *et al.* (2013)].

6. Experimental results and discussion

The two-dimensional experimental distribution of the polarization of NANOPERM is depicted in Figs. 6 $(P_{f\perp}^+)$ and 7 $(P_{f\perp}^-)$



Figure 6

Qualitative comparison between experiment and theory. (a)–(d) Two-dimensional experimental polarization $P_{f\perp}^+(\mathbf{q})$ of the scattered neutrons of NANOPERM [(Fe_{0.985}Co_{0.015})₉₀Zr₇B₃] at a series of applied magnetic fields (see insets). **H**₀ is horizontal in the plane. The range of momentum transfers is restricted to $q \leq 0.33 \text{ nm}^{-1}$. (e)–(h) Prediction by the analytical micromagnetic theory (no free parameters) using the experimental ratio $\alpha_{exp}(q)$ [equation (63)] and the structural ($\xi_{\rm M} = \xi_{\rm H} = D/2 = 7.5 \text{ nm}$) and magnetic (A, M₀) interaction parameters of NANOPERM [see text, Michels *et al.* (2012) and Honecker *et al.* (2013)]. The central white octagons mark the position of the beamstop.





Figure 8

(Data points) Experimental polarizations $P_{f\perp}^+(q, \theta = 90^\circ)$ (a) and $P_{f\perp}^-(q, \theta = 90^\circ)$ (b) of the scattered neutrons of NANOPERM [(Fe_{0.985}Co_{0.015})₉₀Zr₇B₃] at a series of internal magnetic fields (see inset). For clarity of presentation, error bars are only shown for one field. (Solid lines) Prediction by the analytical micromagnetic theory [equations (44a) and (44b)] using the ratio $\alpha_{exp}(q)$ [equation (63)]. Note the different scales on the ordinates in (a) and (b).

at selected field values together with a qualitative comparison with the simulated polarization based on the micromagnetic SANS theory [see Michels et al. (2012) for some selected spinresolved SANS cross sections]. The theory uses as input values the experimental ratio $\alpha_{exp}(q)$ [equation (63)] and the structural ($\xi_{\rm M} = \xi_{\rm H} = D/2 = 7.5$ nm) and magnetic (A, M₀) interaction parameters. In agreement with the previous micromagnetic SANS data analysis of this sample (Michels et al., 2012; Honecker et al., 2013), we have set the ratio $A_{\rm H}/A_{\rm M} = 0.2$. We also assumed that both spin-flip channels are equal, *i.e.* $d\Sigma^{+-}/d\Omega = d\Sigma^{-+}/d\Omega$, a constraint that was already imposed during the spin-leakage correction. The overall qualitative agreement between experiment and theory (no free parameters) is evident, although the angular anisotropy of the data does not exhibit a large variation with field. Only at the smallest momentum transfers can one notice a change in the anisotropy with decreasing field (in particular in P_{fl}^{-}), which is related to the emerging spin-misalignment scattering; compare e.g. scattering terms $\propto |\widetilde{M}_{v}|^{2} \cos^{4} \theta$ and $\propto CT_{v_7} \sin \theta \cos \theta$ in equations (66) and (69). We also note the existence of (seemingly isotropic) scattering contributions at small $q \lesssim 0.1 \text{ nm}^{-1}$ (especially at 1.27 T), which are probably due to large-scale structures that are not contained in the micromagnetic theory [compare Figs. 6(a) and 6(e) and Figs. 7(a) and 7(e)].

Due to the relatively large statistical noise in the twodimensional $P_{f\perp}^{\pm}$ maps we did not fit the experimental data directly to the theoretical expressions. Therefore, in the following, we consider one-dimensional experimental polarization data, which were obtained by averaging the twodimensional polarized SANS cross sections over $\pm 8^{\circ}$ along the vertical direction ($\theta = 90^{\circ}$). These averages were used in equations (23*a*) and (23*b*) to obtain $P_{f\perp}^{\pm}(q)$. The resulting data in Fig. 8 were then fitted using the general equations (44*a*) and (44*b*) (also averaged over $\pm 8^{\circ}$ along $\theta = 90^{\circ}$). Adjustable parameters are the amplitudes (scaling parameters) $A_{\rm M}$ and $A_{\rm H}$ of, respectively, \tilde{M}_z^2 and \tilde{H}_p^2 as well as the corresponding correlation lengths $\xi_{\rm M}$ and $\xi_{\rm H}$ [compare equations (41) and (42)]. The field-dependent micromagnetic exchange length $l_{\rm H}$, which is contained in the dimensionless function $p(q, H_i)$ [compare equations (38)–(40)], is computed at each field using the materials parameters A and M_0 ; A is treated here as an additional adjustable parameter. For $\alpha(q)$ we used equation (63), and the DMI has been ignored in the data analysis $(l_D = 0)$. Since the $P_{f\perp}^{\pm}$ differ only by the \widetilde{NM}_y and \widetilde{NM}_z interference terms, we have fitted the $P_{f\perp}^{\pm}(q)$ data corresponding to the same field simultaneously. The applied field H_0 has been corrected for demagnetizing effects.

The fits in Fig. 8 (solid lines) provide a reasonable description of the experimental data. The obtained values for $\xi_{\rm M}$ and $\xi_{\rm H}$ are shown in Fig. 9; $\xi_{\rm H} \cong 6-15$ nm is at all fields consistently of the order of the particle size, while $\xi_{\rm M}$ takes on larger values between about 22 and 65 nm. For the exchange-stiffness constant, we obtain (from the four local fits) best-fit values in the range $A = (4.8-9.7) \times 10^{-12}$ J m⁻¹. These values agree very well with data in the literature (Honecker *et al.*, 2013; Bersweiler *et al.*, 2022).

Clearly, more experiments are needed in order to further scrutinize the predictions of the present micromagnetic theory for the uniaxial polarization analysis of bulk ferromagnets. In



Figure 9

Resulting best-fit values for the correlation lengths $\xi_{\rm M}$ and $\xi_{\rm H}$ (see inset). Lines are a guide to the eye.

this respect, the development of computational tools to directly analyze the two-dimensional polarization maps using different form-factor and structure-factor expressions for \widetilde{M}_z^2 and \widetilde{H}_p^2 , and possibly the inclusion of a particle-size distribution function, would be desirable. Likewise, SANS measurements at a preferably saturating magnetic field are necessary to determine the nuclear SANS cross section, *e.g.* via a horizontal average of the non-spin-flip SANS cross section.

7. Summary and conclusions

We have provided a micromagnetic theory for the uniaxial polarization of the scattered neutrons of bulk ferromagnets, as it can be measured by means of the small-angle neutron scattering (SANS) method. The theoretical expressions contain the effects of an isotropic exchange interaction, the Dzyaloshinskii-Moriya interaction, magnetic anisotropy, magnetodipolar interaction and an external magnetic field. The theory has been employed to analyze experimental data on a soft magnetic nanocrystalline alloy; it may provide information on the magnetic interactions (exchange and DMI constants) and on the spatial structures of the magnetic anisotropy and magnetostatic fields. Given that uniaxial polarization analysis is becoming more and more available on SANS instruments worldwide and in view of the recent seminal progress made regarding several techniques which exploit the neutron polarization degree of freedom to characterize large-scale magnetic structures (SESANS, DFI, SEMSANS), we believe that the results of this paper open up a new avenue for magnetic neutron data analysis on mesoscopic magnetic systems. This is because the presented micromagnetic SANS framework forms the basis for all of these new and promising polarization encoding techniques, with the paper by Rekveldt et al. (2006) providing the relevant expressions that link the magnetization distribution to the final polarization and the projected correlation function.

APPENDIX A Non-spin-flip, spin-flip and SANSPOL cross sections

In this appendix, we present the expressions for the polarized SANS cross sections in terms of the Fourier components $\widetilde{M}_{x,y,z}(\mathbf{q})$ of the magnetization. The two non-spin-flip and the two spin-flip POLARIS cross sections carry, respectively, the superscripts ++ and -- and +- and -+, and the subscripts \perp and \parallel refer to the respective scattering geometry (compare Fig. 1) (Michels, 2014):

$$\frac{\mathrm{d}\Sigma_{\perp}^{\pm\pm}}{\mathrm{d}\Omega} = K \Big(b_{\mathrm{H}}^{-2} |\widetilde{N}|^2 + |\widetilde{M}_{y}|^2 \sin^2\theta \cos^2\theta + |\widetilde{M}_{z}|^2 \sin^4\theta \\ - \mathrm{CT}_{yz} \sin^3\theta \cos\theta \mp b_{\mathrm{H}}^{-1} \mathrm{CT}_{\widetilde{NM}_{z}} \sin^2\theta \\ \pm b_{\mathrm{H}}^{-1} \mathrm{CT}_{\widetilde{NM}_{y}} \sin\theta \cos\theta \Big), \tag{64}$$

$$\frac{\mathrm{d}\Sigma_{\parallel}^{\pm\pm}}{\mathrm{d}\Omega} = K \bigg(b_{\mathrm{H}}^{-2} |\widetilde{N}|^2 + |\widetilde{M}_z|^2 \mp b_{\mathrm{H}}^{-1} \mathrm{CT}_{\widetilde{NM}_z} \bigg), \qquad (65)$$

$$\frac{\mathrm{d}\Sigma_{\perp}^{\pm\mp}}{\mathrm{d}\Omega} = K \Big(|\widetilde{M}_{x}|^{2} + |\widetilde{M}_{y}|^{2} \cos^{4}\theta + |\widetilde{M}_{z}|^{2} \sin^{2}\theta \cos^{2}\theta - \mathrm{CT}_{yz} \sin\theta \cos^{3}\theta \mp i\chi \Big), \tag{66}$$

$$\frac{\mathrm{d}\Sigma_{\parallel}^{\pm\mp}}{\mathrm{d}\Omega} = K\Big(|\widetilde{M}_{x}|^{2}\sin^{2}\theta + |\widetilde{M}_{y}|^{2}\cos^{2}\theta - \mathrm{CT}_{xy}\sin\theta\cos\theta\Big),\tag{67}$$

where $K = 8\pi^3 V^{-1} b_{\rm H}^2$, and the chiral function $\chi(\mathbf{q})$ is given by

$$\chi = \left(\widetilde{M}_{x}\widetilde{M}_{y}^{*} - \widetilde{M}_{x}^{*}\widetilde{M}_{y}\right)\cos^{2}\theta - \left(\widetilde{M}_{x}\widetilde{M}_{z}^{*} - \widetilde{M}_{x}^{*}\widetilde{M}_{z}\right)\sin\theta\cos\theta.$$
(68)

Note that $\chi(\textbf{q})=0$ for $\textbf{k}_0 \parallel \textbf{H}_0$. The two SANSPOL cross sections $d\Sigma^+/d\Omega$ = $d\Sigma^{++}/d\Omega$ + $d\Sigma^{+-}/d\Omega$ and $d\Sigma^-/d\Omega$ = $d\Sigma^{--}/d\Omega$ + $d\Sigma^{-+}/d\Omega$ read

$$\frac{\mathrm{d}\Sigma_{\perp}^{\pm}}{\mathrm{d}\Omega} = K \Big(b_{\mathrm{H}}^{-2} |\widetilde{N}|^{2} + |\widetilde{M}_{x}|^{2} + |\widetilde{M}_{y}|^{2} \cos^{2}\theta \\ + |\widetilde{M}_{z}|^{2} \sin^{2}\theta - \mathrm{CT}_{yz} \sin\theta\cos\theta \\ \mp b_{\mathrm{H}}^{-1} \mathrm{CT}_{\widetilde{N}\widetilde{M}_{z}} \sin^{2}\theta \pm b_{\mathrm{H}}^{-1} \mathrm{CT}_{\widetilde{N}\widetilde{M}_{y}} \sin\theta\cos\theta \mp i\chi \Big), \quad (69)$$

$$\frac{\mathrm{d}\Sigma_{\parallel}^{\pm}}{\mathrm{d}\Omega} = K \Big(b_{\mathrm{H}}^{-2} |\widetilde{N}|^{2} + |\widetilde{M}_{x}|^{2} \sin^{2}\theta + |\widetilde{M}_{y}|^{2} \cos^{2}\theta \\ + |\widetilde{M}_{z}|^{2} - \mathrm{CT}_{xy} \sin\theta \cos\theta \mp b_{\mathrm{H}}^{-1} \mathrm{CT}_{\widetilde{N}\widetilde{M}_{z}} \Big).$$
(70)

The magnetic-magnetic and nuclear-magnetic cross terms have been abbreviated as follows:

$$CT_{yz} = \widetilde{M}_{y}\widetilde{M}_{z}^{*} + \widetilde{M}_{y}^{*}\widetilde{M}_{z}, \qquad (71a)$$

$$CT_{xy} = \widetilde{M}_x \widetilde{M}_y^* + \widetilde{M}_x^* \widetilde{M}_y, \qquad (71b)$$

$$\mathrm{CT}_{\widetilde{NM_z}} = \widetilde{N}\widetilde{M}_z^* + \widetilde{N}^*\widetilde{M}_z, \qquad (71c)$$

$$\mathrm{CT}_{\widetilde{NM}_{y}} = \widetilde{N}\widetilde{M}_{y}^{*} + \widetilde{N}^{*}\widetilde{M}_{y}. \tag{71d}$$

In actual SANSPOL and POLARIS experiments the neutron optics (polarizer, spin flipper, analyzer) do not work perfectly and polarization corrections become necessary. The incident beam polarization efficiency may be denoted by $p^+ = I^+/(I^+ + I^-)$, where I^\pm are, respectively, the number of neutrons with spins aligned antiparallel and parallel with respect to \mathbf{H}_0 ; note that $p^+ = 1/2$ for an unpolarized beam. The efficiency of the spin flipper is ϵ^\pm with $\epsilon^+ = 0$ for flipper off and $\epsilon^- = \epsilon \cong 1$ for flipper on. We emphasize that the half-polarized SANSPOL cross sections $d\Sigma^+/d\Omega$ and $d\Sigma^-/d\Omega$ can be obtained directly and corrected for nonideal neutron polarization provided that the parameters p^+ and ϵ are known from reference measurements. For POLARIS, it is necessary to measure all four partial cross sections $d\Sigma^{++}/d\Omega$, $d\Sigma^{--}/d\Omega$, $d\Sigma^{+-}/d\Omega$ and $d\Sigma^{-+}/d\Omega$ in order to correct for spin leakage



Figure 10 Plot of $P_{f\perp}^+(q_y, q_z)$ (upper row) and $P_{f\perp}^-(q_y, q_z)$ (lower row) for different applied magnetic fields H_i (see insets). $\alpha = \alpha(q)$ [equation (63)], $A_H/A_M = 0.5$, $l_D = 0$.

between the different channels (Wildes, 2006). Such corrections can *e.g.* be accomplished by means of the *BerSANS* (Keiderling, 2002; Keiderling *et al.*, 2008), *Pol-Corr* (Krycka *et al.*, 2012) and *GRASP* (Dewhurst, 2021) software tools.

Moreover, we note that $d\Sigma^{+-}/d\Omega = d\Sigma^{-+}/d\Omega$ for many polycrystalline bulk ferromagnets (Honecker *et al.*, 2010). However, in our theoretical treatment we explicitly take into account the polarization dependence of the SANSPOL and spin-flip cross sections via the chiral function $\chi(\mathbf{q})$. This is relevant *e.g.* for systems where inversion symmetry is broken and the DMI is operative (Michels *et al.*, 2019; Quan *et al.*, 2020).

APPENDIX **B**

Selected results for the polarization of the scattered neutrons of bulk ferromagnets

In this appendix we provide some selected graphical representations for the dependency of the polarization of the scattered neutrons on the magnitude and orientation of the scattering vector, the applied magnetic field, the ratio of $A_{\rm M}$ to $A_{\rm H}$, the ratio of nuclear to longitudinal magnetic scattering, and the DMI (via the exchange length $l_{\rm D}$). Only results for $\mathbf{k}_0 \perp \mathbf{H}_0$ are shown. The following materials parameters are used: $A = 4.7 \text{ pJ m}^{-1}$; $\mu_0 M_0 = 1.64 \text{ T}$ ($l_{\rm M} \cong 2.1 \text{ nm}$); $D = 2 \text{ mJ m}^{-2}$ ($l_{\rm D} \cong 1.9 \text{ nm}$) (Honecker *et al.*, 2013).

Fig. 10 shows the two-dimensional final polarization $P_{\rm fl}^{\pm}(q_{\rm v},q_{\rm z})$ and Fig. 11 depicts the corresponding 2π azimuthally averaged data $P_{f\perp}^{\pm}(q)$ for different values of the applied magnetic field H_i , $\alpha = \alpha(q)$ [equation (63)], $A_{\rm H}/A_{\rm M} = 0.5, l_{\rm D} = 0$. The local extrema in $P_{\rm f\perp}^{\pm}(q)$ at small q are due to $\alpha_{exp}(q)$; setting α = constant results in smooth and continuously decaying functions. Figs. 12 and 13 display the polarization $P^{\pm}(\mathbf{q})$ for different ratios of $A_{\rm H}/A_{\rm M}$ (Fig. 12) and for different (constant) α values (Fig. 13) at a constant field of $\mu_0 H_i = 0.3 \text{ T}$ and for $l_D = 0$. Including the DMI results in asymmetric P^{\pm} patterns at nonsaturating fields (see Fig. 14). For the calculation of P^{\pm} according to equation (44*a*), we have up to now assumed Lorentzian-squared functions for $\widetilde{M}_z^2(q\xi_M)$ and $\widetilde{H}_{p}^{2}(q\xi_{\rm H})$ [compare equations (41)–(42)] with $\xi_{\rm M} = \xi_{\rm H_{a}}$ D/2 = 7.5 nm. The effect of a hard-sphere form factor for \tilde{M}_z^2 and \widetilde{H}_{p}^{2} with different values for ξ_{M} and ξ_{H} is depicted in Fig. 15. Here, peak-type features may appear in P^{\pm} , which might be detected in highly monodisperse particulate systems.



Figure 11 2^{-1} simulation $R^{+}(z)(z)$ and $R^{+}(z)(z)$

 2π azimuthally averaged $P_{f\perp}^+(q)$ (a) and $P_{f\perp}^-(q)$ (b) of the data shown in Fig. 10.



Figure 12 Plot of $P_{f\perp}^+(q_y, q_z)$ (upper row) and $P_{f\perp}^-(q_y, q_z)$ (lower row) for different ratios of A_H/A_M (see insets). $\alpha = \alpha(q)$ [equation (63)], $\mu_0 H_i = 0.3$ T, $l_D = 0$.



Figure 13 Plot of $P_{f\perp}^+(q_y, q_z)$ (upper row) and $P_{f\perp}^-(q_y, q_z)$ (lower row) for different values of α = constant (see insets). $A_{\rm H}/A_{\rm M} = 0.2$, $\mu_0 H_{\rm i} = 0.3$ T, $l_{\rm D} = 0$.



Figure 14

Effect of the DMI. Plot of $P_{f\perp}^+(q_y, q_z)$ (upper row) and $P_{f\perp}^-(q_y, q_z)$ (lower row) as a function of H_i (see insets). $\alpha = \alpha(q)$ [equation (63)], $A_H/A_M = 1$, $l_D = 1.9$ nm.



Figure 15

Results for the azimuthally averaged $P_{f\perp}^+(q)$ using the sphere form factor (instead of Lorentzian-squared functions) for both $\widetilde{M}_z^2(q\xi_M)$ and $\widetilde{H}_p^2(q\xi_H)$. (*a*) Field dependence (see inset) of $P_{f\perp}^+(q)$ for $\xi_M = 7.5$ nm and $\xi_H = 8.5$ nm. (*b*) $P_{f\perp}^+(q)$ at $\mu_0 H_i = 0.3$ T, $\xi_M = 7.5$ nm, but for increasing ξ_H (see inset). (*c*) $P_{f\perp}^+(q)$ at $\mu_0 H_i = 0.3$ T, $\xi_H = 7.5$ nm, but for increasing ξ_M (see inset). $\alpha = \alpha(q)$ [equation (63)], $A_H/A_M = 0.2$, $l_D = 0$.

Acknowledgements

This work is partially based on experiments performed at the Institut Laue–Langevin, Grenoble, France.

Funding information

Andreas Michels and Artem Malyeyev thank the National Research Fund of Luxembourg for financial support (AFR grant No. 12417141).

References

- Aharoni, A. (2000). Introduction to the Theory of Ferromagnetism, 2nd ed. Oxford University Press.
- Aswal, V. K., van den Brandt, B., Hautle, P., Kohlbrecher, J., Konter, J. A., Michels, A., Piegsa, F. M., Stahn, J., Van Petegem, S. & Zimmer, O. (2008). Nucl. Instrum. Methods Phys. Res. A, 586, 86–89.
- Batz, M., Baessler, S., Heil, W., Otten, E. W., Rudersdorf, D., Schmiedeskamp, J., Sobolev, Y. & Wolf, M. (2005). J. Res. Natl Inst. Stand. Technol. 110, 293.
- Benacchio, G., Titov, I., Malyeyev, A., Peral, I., Bersweiler, M., Bender, P., Mettus, D., Honecker, D., Gilbert, E. P., Coduri, M., Heinemann, A., Mühlbauer, S., Çakır, A., Acet, M. & Michels, A. (2019). *Phys. Rev. B*, **99**, 184422.
- Bender, P., Fock, J., Frandsen, C., Hansen, M. F., Balceris, C., Ludwig, F., Posth, O., Wetterskog, E., Bogart, L. K., Southern, P., Szczerba, W., Zeng, L., Witte, K., Grüttner, C., Westphal, F., Honecker, D., González-Alonso, D., Fernández Barquín, L. & Johansson, C. (2018). J. Phys. Chem. C, 122, 3068–3077.
- Bender, P., Honecker, D. & Fernández Barquín, L. (2019). Appl. Phys. Lett. 115, 132406.
- Bender, P., Wetterskog, E., Honecker, D., Fock, J., Frandsen, C., Moerland, C., Bogart, L. K., Posth, O., Szczerba, W., Gavilán, H., Costo, R., Fernández-Díaz, M. T., González-Alonso, D., Fernández Barquín, L. & Johansson, C. (2018). *Phys. Rev. B*, 98, 224420.
- Bersweiler, M., Adams, M. P., Peral, I., Kohlbrecher, J., Suzuki, K. & Michels, A. (2022). *IUCrJ*, 9, 65–72.
- Bloch, F. (1936). Phys. Rev. 50, 259-260.
- Bloch, F. (1937). Phys. Rev. 51, 994.
- Blume, M. (1963). Phys. Rev. 130, 1670-1676.
- Brandt, B. van den, Glättli, H., Grillo, I., Hautle, P., Jouve, H., Kohlbrecher, J., Konter, J. A., Leymarie, E., Mango, S., May, R. P., Michels, A., Stuhrmann, H. B. & Zimmer, O. (2006). *Eur. Phys. J. B*, 49, 157–165.
- Brown, P. J. (2006). *Neutron Scattering from Magnetic Materials*, edited by T. Chatterji, pp. 215–244. Amsterdam: Elsevier.
- Brown, P. J., Forsyth, J. B. & Tasset, F. (1993). *Proc. R. Soc. London A*, **442**, 147–160.
- Brown, W. F. Jr (1963). *Micromagnetics*. New York: Interscience Publishers.
- Burgy, M., Hughes, D. J., Wallace, J. R., Heller, R. B. & Woolf, W. E. (1950). *Phys. Rev.* **80**, 953–960.
- Chatterji, T. (2006). *Neutron Scattering from Magnetic Materials*. Amsterdam: Elsevier.
- Dewhurst, C. D. (2021). Graphical Reduction and Analysis SANS Program (GRASP). Institut Laue-Langevin, Grenoble, France.
- Drabkin, G. M., Okorokov, A. I. & Runov, V. V. (1972). JETP Lett. 15, 324.
- Dufour, C., Fitzsimmons, M. R., Borchers, J. A., Laver, M., Krycka, K. L., Dumesnil, K., Watson, S. M., Chen, W. C., Won, J. & Singh, S. (2011). *Phys. Rev. B*, **84**, 064420.
- Grigoriev, S. V., Kraan, W. H., Rekveldt, M. Th., Kruglov, T. & Bouwman, W. G. (2006). J. Appl. Cryst. 39, 252–258.
- Grigoriev, S. V., Sukhanov, A. S., Altynbaev, E. V., Siegfried, S., Heinemann, A., Kizhe, P. & Maleyev, S. V. (2015). *Phys. Rev. B*, 92, 220415.
- Grutter, A. J., Krycka, K. L., Tartakovskaya, E. V., Borchers, J. A., Reddy, K. S. M., Ortega, E., Ponce, A. & Stadler, B. J. H. (2017). *ACS Nano*, **11**, 8311–8319.
- Halpern, O. & Holstein, T. (1941). Phys. Rev. 59, 960-981.
- Halpern, O. & Johnson, M. H. (1939). Phys. Rev. 55, 898-923.
- Hasz, K., Ijiri, Y., Krycka, K. L., Borchers, J. A., Booth, R. A., Oberdick, S. & Majetich, S. A. (2014). *Phys. Rev. B*, 90, 180405.
- Honecker, D., Dewhurst, C. D., Suzuki, K., Erokhin, S. & Michels, A. (2013). *Phys. Rev. B*, **88**, 094428.
- Honecker, D., Ferdinand, A., Döbrich, F., Dewhurst, C. D., Wiedenmann, A., Gómez-Polo, C., Suzuki, K. & Michels, A. (2010). Eur. Phys. J. B, 76, 209–213.

- Honecker, D., Fernández Barquín, L. & Bender, P. (2020). *Phys. Rev. B*, **101**, 134401.
- Honecker, D. & Michels, A. (2013). Phys. Rev. B, 87, 224426.
- Hughes, D. J., Burgy, M. T., Heller, R. B. & Wallace, J. W. (1949). *Phys. Rev.* **75**, 565–569.
- Hughes, D. J., Wallace, J. R. & Holtzman, R. H. (1948). *Phys. Rev.* 73, 1277–1290.
- Ijiri, Y., Krycka, K. L., Hunt-Isaak, I., Pan, H., Hsieh, J., Borchers, J. A., Rhyne, J. J., Oberdick, S. D., Abdelgawad, A. & Majetich, S. A. (2019). *Phys. Rev. B*, **99**, 094421.
- Ito, N., Michels, A., Kohlbrecher, J., Garitaonandia, J. S., Suzuki, K. & Cashion, J. D. (2007). *J. Magn. Magn. Mater.* **316**, 458–461.
- Keiderling, U. (2002). Appl. Phys. Mater. Sci. Process. 74, s1455s1457.
- Keiderling, U., Wiedenmann, A., Rupp, A., Klenke, J. & Heil, W. (2008). Meas. Sci. Technol. 19, 034009.
- Kronmüller, H. & Fähnle, M. (2003). Micromagnetism and the Microstructure of Ferromagnetic Solids. Cambridge University Press.
- Krycka, K., Chen, W., Borchers, J., Maranville, B. & Watson, S. (2012). J. Appl. Cryst. 45, 546–553.
- Krycka, K. L., Booth, R. A., Hogg, C. R., Ijiri, Y., Borchers, J. A., Chen, W. C., Watson, S. M., Laver, M., Gentile, T. R., Dedon, L. R., Harris, S., Rhyne, J. J. & Majetich, S. A. (2010). *Phys. Rev. Lett.* **104**, 207203.
- Krycka, K. L., Borchers, J. A., Booth, R. A., Ijiri, Y., Hasz, K., Rhyne, J. J. & Majetich, S. A. (2014). *Phys. Rev. Lett.* **113**, 147203.
- Laver, M., Mudivarthi, C., Cullen, J. R., Flatau, A. B., Chen, W. C., Watson, S. M. & Wuttig, M. (2010). *Phys. Rev. Lett.* **105**, 027202.
- Li, F., Steinke, N.-J., Dalgliesh, R. M., Washington, A. L., Shen, J., Pynn, R. & Parnell, S. R. (2021). *Nucl. Instrum. Methods Phys. Res. A*, **1014**, 165705.
- Lovesey, S. W. (1984). Theory of Neutron Scattering from Condensed Matter, Vols. I and II. Oxford: Clarendon Press.
- Maleev, S. V. (2002). Phys.-Usp. 45, 569-596.
- Maleev, S. V., Bar'yakhtar, V. G. & Suris, R. A. (1963). Sov. Phys. Solid State, 4, 2533.
- Maleev, S. V. & Ruban, V. A. (1970). Sov. Phys. JETP, 31, 111-116.
- Maleev, S. V. & Ruban, V. A. (1972). Sov. Phys. JETP, 35, 222-226.
- Metlov, K. L. & Michels, A. (2015). Phys. Rev. B, 91, 054404.
- Mettus, D. & Michels, A. (2015). J. Appl. Cryst. 48, 1437–1450.
- Mezei, F. (1986). Phys. B+C, 137, 295-308.
- Michels, A. (2014). J. Phys. Condens. Matter, 26, 383201.
- Michels, A. (2021). Magnetic Small-Angle Neutron Scattering. A Probe for Mesoscale Magnetism Analysis. Oxford University Press.
- Michels, A., Honecker, D., Döbrich, F., Dewhurst, C. D., Suzuki, K. & Heinemann, A. (2012). *Phys. Rev. B*, 85, 184417.
- Michels, A., Malyeyev, A., Titov, I., Honecker, D., Cubitt, R., Blackburn, E. & Suzuki, K. (2020). *IUCrJ*, **7**, 136–142.
- Michels, A., Mettus, D., Honecker, D. & Metlov, K. L. (2016). *Phys. Rev. B*, **94**, 054424.
- Michels, A., Mettus, D., Titov, I., Malyeyev, A., Bersweiler, M., Bender, P., Peral, I., Birringer, R., Quan, Y., Hautle, P., Kohlbrecher, J., Honecker, D., Fernández, J. R., Barquín, L. F. & Metlov, K. L. (2019). *Phys. Rev. B*, **99**, 014416.
- Moon, R. M., Riste, T. & Koehler, W. C. (1969). Phys. Rev. 181, 920-931.
- Mudivarthi, C., Laver, M., Cullen, J., Flatau, A. B. & Wuttig, M. (2010). J. Appl. Phys. 107, 09A957.
- Niketic, N., van den Brandt, B., Wenckebach, W. Th., Kohlbrecher, J. & Hautle, P. (2015). J. Appl. Cryst. 48, 1514–1521.
- Noda, Y., Koizumi, S., Masui, T., Mashita, R., Kishimoto, H., Yamaguchi, D., Kumada, T., Takata, S., Ohishi, K. & Suzuki, J. (2016). J. Appl. Cryst. **49**, 2036–2045.
- Oberdick, S. D., Abdelgawad, A., Moya, C., Mesbahi-Vasey, S., Kepaptsoglou, D., Lazarov, V. K., Evans, R. F. L., Meilak, D., Skoropata, E., van Lierop, J., Hunt-Isaak, I., Pan, H., Ijiri, Y., Krycka, K. L., Borchers, J. A. & Majetich, S. A. (2018). *Sci. Rep.* 8, 3425.

- Okorokov, A. I., Gukasov, A. G., Runov, V. V., Mikhailova, V. E. & Roth, M. (1981). Sov. Phys. JETP, **54**, 775.
- Okorokov, A. I. & Runov, V. V. (2001). Physica B, 297, 239-244.
- Okorokov, A. I., Runov, V. V. & Gukasov, A. G. (1978). Nucl. Instrum. Methods, 157, 487–493.
- Okudaira, T., Oku, T., Ino, T., Hayashida, H., Kira, H., Sakai, K., Hiroi, K., Takahashi, S., Aizawa, K., Endo, H., Endo, S., Hino, M., Hirota, K., Honda, T., Ikeda, K., Kakurai, K., Kambara, W., Kitaguchi, M., Oda, T., Ohshita, H., Otomo, T., Shimizu, H. M., Shinohara, T., Suzuki, J. & Yamamoto, T. (2020). *Nucl. Instrum. Methods Phys. Res. A*, **977**, 164301.
- Orue, I., Marcano, L., Bender, P., García-Prieto, A., Valencia, S., Mawass, M. A., Gil-Cartón, D., Alba Venero, D., Honecker, D., García-Arribas, A., Fernández Barquín, L., Muela, A. & Fdez-Gubieda, M. L. (2018). *Nanoscale*, **10**, 7407–7419.
- Petoukhov, A. K., Guillard, V., Andersen, K. H., Bourgeat-Lami, E., Chung, R., Humblot, H., Jullien, D., Lelievre-Berna, E., Soldner, T., Tasset, F. & Thomas, M. (2006). *Nucl. Instrum. Methods Phys. Res.* A, 560, 480–484.
- Pynn, R., Hayter, J. B. & Charles, S. W. (1983). *Phys. Rev. Lett.* **51**, 710–713.
- Quan, Y., van den Brandt, B., Kohlbrecher, J. & Hautle, P. (2019*a*). J. *Phys. Conf. Ser.* **1316**, 012010.
- Quan, Y., van den Brandt, B., Kohlbrecher, J., Wenckebach, W. Th. & Hautle, P. (2019b). Nucl. Instrum. Methods Phys. Res. A, 921, 22–26.
- Quan, Y., Kohlbrecher, J., Hautle, P. & Michels, A. (2020). J. Phys. Condens. Matter, **32**, 285804.
- Ramazanoglu, M., Laver, M., Ratcliff, W., Watson, S. M., Chen, W. C., Jackson, A., Kothapalli, K., Lee, S., Cheong, S. W. & Kiryukhin, V. (2011). *Phys. Rev. Lett.* **107**, 207206.
- Rekveldt, M. Th. (1971). J. Phys. Colloq. C1, 32, 579.
- Rekveldt, M. Th. (1973). Z. Phys. 259, 391-410.
- Rekveldt, M. T., van Dijk, N. H., Grigoriev, S. V., Kraan, W. H. & Bouwman, W. G. (2006). *Rev. Sci. Instrum.* 77, 073902.
- Rosman, R. & Rekveldt, M. Th. (1991). Phys. Rev. B, 43, 8437-8449.
- Schärpf, O. & Capellmann, H. (1993). Phys. Status Solidi A, 135, 359–379.
- Schweizer, J. (2006). Neutron Scattering from Magnetic Materials, edited by T. Chatterji, pp. 153–213. Amsterdam: Elsevier.
- Schwinger, J. S. (1937). Phys. Rev. 51, 544-552.
- Shull, C. G., Wollan, E. O. & Koehler, W. C. (1951). Phys. Rev. 84, 912–921.
- Squires, G. L. (2012). Introduction to the Theory of Thermal Neutron Scattering. Cambridge University Press.
- Stewart, J. R., Giblin, S. R., Honecker, D., Fouquet, P., Prabhakaran, D. & Taylor, J. W. (2019). J. Phys. Condens. Matter, 31, 025802.
- Suzuki, K. & Herzer, G. (2006). Advanced Magnetic Nanostructures, edited by D. Sellmyer & R. Skomski, pp. 365–401. New York: Springer-Verlag.
- Suzuki, K., Makino, A., Inoue, A. & Masumoto, T. (1994). J. Magn. Soc. Jpn, 18, 800–804.
- Takagi, R., White, J. S., Hayami, S., Arita, R., Honecker, D., Rønnow, H. M., Tokura, Y. & Seki, S. (2018). Sci. Adv. 4, eaau3402.
- Tasset, F. (1989). Physica B, 156–157, 627–630.
- Tasset, F., Brown, P. J., Lelièvre-Berna, E., Roberts, T., Pujol, S., Allibon, J. & Bourgeat-Lami, E. (1999). *Physica B*, 267–268, 69–74.
- Thoma, H., Hutanu, V., Deng, H., Dmitrienko, V., Brown, P., Gukasov, A., Roth, G. & Angst, M. (2021). *Phys. Rev. X*, **11**, 011060.
- Ueland, B. G., Lynn, J. W., Laver, M., Choi, Y. J. & Cheong, S. W. (2010). *Phys. Rev. Lett.* **104**, 147204.
- Valsecchi, J., Kim, Y., Lee, S. W., Saito, K., Grünzweig, C. & Strobl, M. (2021). *Sci. Rep.* **11**, 8023.
- White, J. S., Butykai, Á., Cubitt, R., Honecker, D., Dewhurst, C. D., Kiss, L. F., Tsurkan, V. & Bordács, S. (2018). *Phys. Rev. B*, 97, 020401.
- Wiedenmann, A. (2005). Physica B, 356, 246-253.
- Wildes, A. R. (2006). Neutron News, 17(2), 17-25.
- Williams, W. G. (1988). Polarized Neutrons. Oxford: Clarendon Press.